Reculations	SROW	This space for use	by IRRC			
(1) Agency	A CONTRACTOR OF THE PROPERTY O	Bush	!			
Department of Conservation and Natura	l Resources		•			
		90,607-3 // /043	2.			
(2) I.D. Number (Governor's Office Use)	· - .	A SOTO O LO COMO SEON	et :			
7B-1		IRRC Number:	1887.			
(3) Short Title			000			
State Forest Rules and Regulations			•			
(4) PA Code Cite (5) Agend	cy Contacts & Tel	ephone Numbers				
21 Pa Code Chapter 21 Prima	ry Contact: Warn	en A. Ely 717-787-	2014			
Secon	dary Contact: H.	E. Kelly 717-787-	2015			
(6) Type of Rulemaking (Check One)	(7) Is a 120- Attached?	Day Emergency Certif	ication			
Proposed RulemakingFinal Order Adopting RegulationFinal Order, Proposed Rulemaking Omitt	_x_No Yes: B	y the Attorney General y the Governor				
(8) Briefly explain the regulation in clear and n	ontechnical langu	age.				
It has not been revised for more than and regulations are badly outdated and Department to protect and control the and commercial use of State Forest lar	Chapter 21 governs the conduct of persons visiting or using State Forest land. It has not been revised for more than 25 years. In their current form, the rules and regulations are badly outdated and are no longer sufficient to allow the Department to protect and control the use of State Forest land. The recreation and commercial use of State Forest land has increased dramatically in the last quarter century and these revisions are sorely needed.					
•						
(9) State the statutory authority for the regulation	n and any relevan	t state or federal court	decisions.			
The Conservation and Natural Resources Code, 18 Pa C.S. §§7505 and 7506, unle relevant state or federal court decision	ess otherwise r					

Regulatory Analysis Form

(10) Is the regulation mandated by any federal or state law or court order, or federal regulation? If yes, cite the specific law, case or regulation, and any deadlines for action.

Yes. The Department is required by Section 313 of the Conservation and Natural Resource Act, 71 P.S. §1340.313, to promulgate regulations for the control, management, protection, utilization, development and use of the lands and resources of State Forests as it may deem necessary or proper to conserve the interests of the Commonwealth, and as necessary to carry out the act.

(11) Explain the compelling public interest that justifies the regulation. What is the problem it addresses?

The rules & regulations which govern conduct on State Forest land have not been revised for more than 25 years. In their current form, they are badly out-dated and are no longer sufficient to allow the Department to protect and manage the use of State Forest land or to adequately insure the safety of visitors. The recreational and commercial use of State Forest land has increased dramatically over the last quarter century and these revisions are sorely needed.

(12) State the public health, safety, environmental or general welfare risks associated with non-regulation.

Non-regulation would lead to increased risk to State Forest visitors as well as pollution and severe damage to the environment.

(13) Describe who will benefit from the regulation. (Quantify the benefits as completely as possible and approximate the number of people who will benefit.)

The revisions to the State Forest Rules and Regulations will provide for improved protection of the forest resource and to the public recreating on State Forest land. Therefore, all citizens of the Commonwealth will benefit from their adoption.

Regulation of the Lysis do the

(14) Describe who will be adversely affected by the regulation. (Quantify the adverse effect as completely as possible and approximate the number of people who will be adversely affected.)

It is not expected that anyone will be adversely affected.

(15) List the persons, groups or entities that will be required to comply with the regulation. (Approximate the number of people who will be required to comply.)

All visitors to State Forest land will be required to comply with the rules and regulations. Examples of visitors include hikers, hunters, snowmobilers, cross-country skiers, campers, equestrians and bird-watchers.

(16) Describe the communications with and inputs from the public in the development and drafting of the regulation. List the persons and/or groups who where involved, if applicable.

The public was invited to submit comments during a 60-day period after publication of the proposed rulemaking on September 13, 1997. In addition, the Department's Advisory Council and the House and Senate Environmental Resources and Energy Committees and the Independent Regulatory Review Commission had the opportunity to comment on the proposed rulemaking. The Department also held 19 public meetings across the Commonwealth from September to November 1997, to explain the proposed rulemaking and to respond to questions. In addition, the Department issued many press releases to inform the public about the proposed rulemaking.

(17) Provide a specific estimate of the cost and/or savings to the regulated community associated with compliance, including any legal, accounting or consulting procedures which may be required.

Since no fees are charged in conjunction with the use of State Forest land, there will be no additional cost to State Forest visitors.

Regulatory Analysis Form	
(18) Provide a specific estimate of the cost and/or savings to local go compliance, including any legal, accounting or consulting procedures	
None	
(19) Provide a specific estimate of the cost and/or savings to state go	overnment associated with the
implementation of the regulation, including any legal, accounting or obe required.	
There will be minor costs associated with revising and/cardboard signs. Those costs should not exceed \$1,000.	or creating weather-proof

Regulatory Analysis Form

(20) In the table below, provide an estimate of the fiscal savings and cost associated with implementation and compliance for the regulated community, local government, and state government for the current year and five subsequent years.

	Current FY	FY +1	FY +2	FY +3	FY +4	FY +5
	Year	Year	Year	Year	Year	Year
SAVINGS:	\$	\$	\$	\$	S	\$
Regulated Community						
Local Government						
State Governments						
Total Savings						
COSTS:						
Regulated Community						
Local Government						
State Governments						
Total Cost						
REVENUE LOSSES:						
Regulated Community						
Local Government						
State Governments						
Total Revenue Losses						

(20a) Explain how the cost estimates listed above were derived.

Except for minor costs associated with producing signs, there will be no costs for the regulated community, local government or state government.

	Regi	ulatory Analysis I	Form	
(20b) Provide the pa	ist three year expend	diture history for progr	rams affected by th	e regulation.
Program	FY-3	FY-2	FY-1	Current FY
	* * *		• • •	
				1
(21) Using the cost-l	benefit information	provided above, expla	in how the benefits	s of the regulation
outweigh the adverse	effects and cost.	_		•
Not applicable.				
•				
(22) Describe the nor	nregulatory alternati	ive considered and the	cost associated wi	th those
alternatives. Provide	the reasons for their	r dismissal.		
Non-regulatory al	ternatives would	not be appropria	- ^	
	66211462762 #0424	Hoe be approprie	ce.	
(23) Describe alterna	tive regulatory sche	emes considered and th	he cost associated v	with those schemes.
Provide the reasons for	or their dismissal.			
Not applicable.				

R	eg	ula	tory	A	nalysis	Fo	rm

(24) Are there any provisions that are more stringent than federal standards? If yes, identify the specific provisions and the compelling Pennsylvania interest that demands stronger regulations.
There are no federal standards.
(25) How does the regulation compare with those of other states? Will the regulation put
Pennsylvania at a competitive disadvantage with other states?
All land management agencies across the country have very similar regulations. Adoption of these revisions will not put Pennsylvania at a competitive dis- advantage.
(26) Will the regulation affect existing or proposed regulations of the promulgating agency or other state agencies? If yes, explain and provide specific citations.
No
(27) Will any public hearings or informational meetings be scheduled? Please provide the dates,
times, and locations, if available.
A series of 19 public meetings were held across the state beginning in September 1997.

				'sis		

(28) Will the regulation change existing reporting, record keeping, or other paperwork requirements? Describe the changes and attach copies of forms or reports which will be required as a result of implementation, if available.
No additional reporting, record keeping or other paperwork will be required.
(29) Please list any special provisions which have been developed to meet the particular needs of affected groups or persons including, but not limited to, minorities, elderly, small businesses, and farmers.
Not applicable.
(30) What is the anticipated effective date of the regulation; the date by which compliance with the
regulation will be required; and the date by which any required permits, licenses or other approvals must be obtained?
August 22, 1998.
(31) Provide the schedule for continual review of the regulation.
The State Forest Rules and Regulations will be reviewed two years after their adoption and every five years thereafter.

FACE SH T FOR FILING DOCUMENTS WITH THE LEGISLATIVE REFERENCE BUREAU

(Pursuant to Commonwealth Documents Law)

98 COT - J. A. M. 3.

ABART CONSTRUCTOR

DO NOT WRITE IN THE SPACE

Copy below is hereby certified to be a true and correct copy of a document issued, prescribed or promulgated by:	Copy below is hereby approved as to form and legality. Executive or Independent Agencies.
Dept. of Conservation and Natural Resources MCENCY)	Sandra W. Stoner
DOCUMENT/FISCAL NOTE NO	GATE OF APPROVAL
on X/ Plu C Otwer	(Deputy General Coursel) -(Chief Coursel, Independent Agency) (Strike inapplicable title)
TITLE: John C. Oliver, Secretary EXECUTIVE OFFICER, CHARMAN OR SECRETARY)	Check if applicable. No Attorney General approval or objection within 30 days after submission.
	Dept. of Dept. of Conservation and Natural Resources MARKET PISCAL MOTE NO

Notice of Final Rulemaking Department of Conservation and Natural Resources

(17 Pa. Code Chapter 21)

State Forest Rules and Regulations

FINAL RULEMAKING STATE FOREST RULES AND REGULATIONS

DEPARTMENT OF CONSERVATION AND NATURAL RESOURCES OCTOBER 1998

Department of Conservation and Natural Resources Notice of Final Rulemaking

PA Code Cite

17 Pa. Code, Chapter 21

Regulation Title

State Forest Rules and Regulations

I.D. Number

7B-1

SECRETARY'S CERTIFICATION

I, John C. Oliver, do hereby certify that I have reviewed this regulation and determined that the regulation is consistent with the principles outlined in Executive Order 1996-1.

NOTICE OF FINAL RULEMAKING

DEPARTMENT OF CONSERVATION AND NATURAL RESOURCES

[17 PA. CODE CH. 21]

State Forests

Order

The Secretary; Department of Conservation and Natural Resources, by this order amends 17 Pa. Code, Chapter 21 (relating to general provisions). The amendments will modernize the State Forest regulations which have not been updated in more than 25 years. The amendments will allow the Department to better manage the use of State Forest land and to protect fragile ecosystems.

A. Effective Date

These proposed amendments will go into effect upon publication in the Pennsylvania Bulletin as final rulemaking.

B. Contact Persons

For further information contact Warren A. Ely, Chief, Division of State Forest Management, P. O. Box 8552, Rachel Carson State Office Building, Harrisburg, PA 17105-8552, (717) 787-2014, or Susan Wood, Assistant Counsel, P. O. Box 8767, Rachel Carson State Office Building, Harrisburg, PA 17105-8767, (717) 772-4171. Persons with a disability may use the AT&T Relay Service by calling (800) 654-5984 (TDD users) or (800) 654-5988 (voice users).

C. Statutory Authority

The final rulemaking is being made under the authority of section 313 of the Conservation and Natural Resources Act (71 P. S. §§ 1340.102–1340.1103) which directs the Department to formulate, adopt and promulgate regulations regarding conduct on State Forest land.

D. Background and Purpose

The regulations which govern conduct on State Forest land have not been revised for more than 25 years. In their current form, they are badly outdated and are no longer sufficient to allow the Department adequately to protect and control the use of State Forest land. The recreational and commercial use of State Forest land has increased dramatically in the last quarter century and these revisions are sorely needed.

The public was invited to submit comments during a 60-day period after publication of the proposed rulemaking on September 13, 1997. In addition, the Department's Advisory Council and the House and Senate Environmental Resources and Energy Committees and the Independent Regulatory Review Commission had the opportunity to comment on the proposed rulemaking. The Department also held 19 public meetings across the Commonwealth from September to November, 1997, to explain the proposed rulemaking and to respond to questions. In addition, the Department issued many press releases to inform the public about the proposed rulemaking.

E. Summary of Comments and Responses on the Proposed Rulemaking

The Department received comments from 1,195 commentators on the proposed rulemaking. Commentators included private citizens, other agencies, legislators and the Independent Regulatory Review Commission.

The following is a discussion of significant comments received.

Sec. 21.1 Definitions

The commentator suggested revising the definition of "commercial activity" to read, "An activity in which a person, organization or association directly or indirectly"

The Department agrees with this suggestion and has changed the wording accordingly.

Sec. 21.1 Definitions

The commentators stated that the definition of "Group Activity" is unclear and ask if group activities include groups of hunters, hikers, etc.

For an activity to fall under the definition of "group activity," three criteria must be met. It must be a scheduled event; it must require a large land area or unique land formation; AND it must have the potential to conflict with traditional uses of State Forest land or pose a potential threat to the environment. Small groups of persons engaged in hunting or hiking would not ordinarily be included in this definition.

Sec. 21.12(b) Mooring and Launching

Section 21.12(b), in its proposed form, would not require launch permits on rivers and streams flowing through State Forest land. The commentator suggested revising the language to include rivers and streams.

The Department agrees. At the time the rulemaking was proposed, there were no launching facilities on rivers and streams on State Forest land. Therefore, the words "lakes and ponds" were substituted for the word "waters." However, since that time, launching facilities have been constructed on State Forest land giving access to the Susquehanna and Juniata Rivers and others are proposed. Therefore, the original word, "waters," has been retained so that the section includes all bodies of water on State Forest land.

Sec. 21.12(d) Mooring and Launching

In proposed rulemaking, the fee schedule for mooring and launching permits appearing in Section 21.12(d) of the current regulations was deleted and replaced with a new provision concerning the launching of boats. The reason for the deletion of the fee schedule was the intention to publish it in the *Pennsylvania Bulletin* rather than promulgating it as a regulation. However, the Department inadvertently omitted any reference in the proposed rulemaking to the fact that fees will continue to be charged for mooring and launching permits and that the fee schedule will be published in the *Pennsylvania Bulletin*. Thus, in final rulemaking, the Department is including this reference to fees in subsection (d). The new provision concerning the launching of boats that had been inserted in proposed subsection (d) is now being moved to subsection (e), and the provision that was in proposed subsection (e) is now subsection (f).

Sec. 21.12(d) (subsection (e) in final rulemaking) Mooring and Launching

The commentator suggested adding wording to Section 21.12(d) (subsection (e) in final rulemaking) to prohibit launching of boats from private docks without the owner's permission.

The Department has not added the suggested provision. The Department does not have the duty or power to enforce private property rights.

Sec. 21.25 Parking

The commentator requested that wording be added to Section 21.25 that would prohibit an able-bodied vehicle operator from parking in designated handicapped parking areas.

The Department agrees with this suggestion. New language has been added accordingly.

Section 21.26(1) Horses and Pack Animals

Section 21.27(1) Bicycles and Person-Powered Vehicles

Many commentators expressed concern with the provisions prohibiting use of horses, pack animals, bicycles and person-powered vehicles on National and State Forest Designated Hiking Trails unless posted as being open to their use. They felt that these provisions unreasonably favor one recreational use (hiking) to the disadvantage of other uses.

Many other commentators supported the provisions, claiming that these other uses, particularly horseback riding, damage trails meant for hiking.

There are two National Hiking Trails and 18 State Forest Designated Hiking Trails in the Commonwealth. The proposed regulations would impose the assumption that these trails would be closed for horses, pack animals, bicycles and person-powered vehicles unless posted as being open. In view of the controversy and confusion generated by these proposed regulations, the Department is deleting them from the proposed rulemaking. The Department intends to revisit these issues and will redraft these regulations regarding the use of horses and mountain bikes on State Forest land in consultation with a work group to be made up of interested groups and individuals. Meanwhile, the Department retains its authority under Section 21.71 to post trails and other areas as being closed to certain uses.

Sec. 21.31(1) Prohibitions

The commentators suggested including a prohibition of planting exotic, invasive plant species. Further, one of the commentators suggested that paragraph (1) be changed by deleting the words, "berries, nuts, leeks and mushrooms" and substituting the words "edible wild plants or plant parts..."

The Department agrees with these comments and has made the suggested revisions.

Sec. 21.61 Camping Permit

The commentator suggested moving the definition of camping to Section 21.1, Definitions.

The Department agrees. The definition has been moved to Section 21.1.

Sec. 21.61 Camping Permit

The commentators objected to Section 21.61 in requiring primitive backpack campers to obtain a camping permit on the grounds that it is overly restrictive and often difficult to conveniently obtain a permit with short notice.

The Department will retain the existing wording in this section so that backpackers may camp without first obtaining a camping permit.

Sec. 21.62 Open fires

The commentators suggested adding a provision at the end of Section 21.62 to the effect that this restriction does not apply to small self-contained camp stoves when used in a safe manner.

The Department agrees with this suggestion and has added the wording.

Sec. 21.62 Open Fires

The commentators object to Section 21.62(b) prohibiting open fires from March 1 to May 25 and from October 1 to December 1

without authorization from the District Forester.

The Department did not make the suggested change. The spring and fall months are the highest fire danger months of the year. The Bureau of Forestry's field staff has been pointing out the need for this prohibition for many years. The District Forester's authorization for an open fire during these months is easy to obtain. A telephone call to a forest district office is all that is necessary since the authorization need not be in writing.

Sec. 21.65 Target Shooting

Since most of the commentators felt this section would allow target shooting virtually anywhere on State Forest land the wording in Section 21.65 is apparently subject to too broad an interpretation.

The Department has changed this section to prohibit target shooting except where authorized by the District Forester.

.

Sec. 21.70a Snow Plowing

The commentators feel that Section 21.70a, which prohibits snow plowing on State Forest roads without the approval of the District Forester, may lead to denying access to property owners.

The suggested change has not been made. The Department does not intend to deny access to any person with a legitimate need. However, it is important that the Department has control over how State Forest roads are plowed. Plowing snow from crowned earthen roads will lead to serious road damage if not done properly. Moreover, many of the State Forest roads are also part of the snowmobile system. It is important that roads be plowed in such a way that this sport is not adversely impacted. However, the wording of the section has been modified so that the District Forester's oral authorization is sufficient.

Sec.21.71 Closure

The commentator suggests that since many provisions of this chapter refer to closure of portions of State Forest land for certain activities or times of year, additional emphasis be placed on this section by moving it to a more prominent location under PRELIMINARY PROVISIONS.

The Department agrees. This section has been relocated and re-numbered as Section 21.4.

Sec. 21.74 Disorderly Conduct

The commentators suggested that a provision be added prohibiting loud noise that might disturb other visitors to State Forest land.

The Department agrees with this comment and has added Paragraph (3) to Section 21.74.

Sec. 21.75 Sanitation

The commentators suggested that this section is too confusing and too restrictive in its proposed form.

The Department agrees with this comment and has re-written the section accordingly.

Sec. 21.76(a) Feeding of Wildlife

The commentators felt that Section 21.76 (a) should be changed to allow the summer feeding of wildlife.

The Department did not make the suggested change. Comments from the Pennsylvania Game Commission, including one from that agency's chief game biologist, state that artificial feeding of wildlife is unnecessary and may actually harm wildlife.

Sec. 21.76 Feeding of Wildlife

The commentators suggested prohibiting artificial feeding of wildlife, except for songbirds. The commentators felt that artificial feeding of wildlife is unnecessary and possibly harmful to wildlife.

The Department did not make this change. Although the Pennsylvania Game Commission discourages artificial feeding of wildlife, many sportsmen have a tradition of feeding wildlife, especially in the winter months. An outright prohibition of this activity would be opposed by a large number of people. Instead, with the Game Commission's assistance, the Department will attempt to educate the public concerning the detrimental aspects of artificial feeding.

Sec. 21.76(b) Feeding of Wildlife

The commentator suggested that the prohibition be against placing of feeders that have a <u>single or combined</u> capacity of more than one-half bushel.

The Department agrees with this comment and has made the suggested change.

Sec. 21.80(b) Trespass

The commentators felt that the reference to 18 Pa. C.S. Section 3503 is vague. There are three subsections in 18 Pa. C.S. Section 3503, each of which describes a different type of trespass and degree of offense. Clarification is needed as to the type of trespass and degree of offense referred to in Section 21.80(b).

The Department has clarified Section 21.80(b) by specifically referring to subsection (b) of 18 Pa. C.S. Section 3503 (relating to defiant trespasser).

Sec. 21.80 Trespass

The commentator objected to this section on the ground that it invites abuse of power.

The Department disagrees. Under the Conservation and Natural Resources Act, the Department has the power and duty to manage, control, protect and regulate the use of State Forest land. The Department, like any landowner, must have the ability to deal with unruly visitors who might cause property damage or who are interfering with the rights of others. The proposed section simply states that a person who engages in this behavior may be ordered by a State Forest officer to leave State Forest land. A person who refuses to leave when ordered commits criminal trespass under Section 3503(b) of the Pennsylvania Crimes Code. The language in the proposed Section 21.80 is identical to that in the Bureau of State Parks rules and regulations.

Simplification, clarification, elimination of redundancies

The following regulations were revised since proposed rulemaking to simplify or clarify their language or to eliminate redundancies. Most of these revisions were made in response to comments. No substantive changes were made to these regulations:

21.11 (Use)

21.12(e) (Mooring and launching)

21.13 (Motorized boats)

21.21(b)(3) (Motorized vehicles)

21.22(b) (Snowmobiles)

21.23(3) (Trail bikes)

21.23a(b) (All-terrain vehicles)

21.23a(c) (All-terrain vehicles)

21.32 (Designated trees)

21.62 (Open fires)

21.63(b) (Hunting and trapping) 21.64(b) (Fishing) 21.76(a) (Feeding of wildlife)

F. Benefits, Costs and Compliance

Benefits

The revisions to the State Forest regulations will provide for improved protection of the forest resource and to the public recreating on State Forest land. Therefore, all citizens will benefit from their adoption.

Compliance Costs

Visitors to State Forest land will be required to comply with the regulations governing conduct. However, there will be no cost to them arising from compliance. There will be a minimal cost to the Department in that it will be necessary to revise some existing signs and create a few new ones.

Compliance Assistance Plan

Not applicable.

Paperwork Requirements

There will be no increase in the amount of required paperwork.

G. Sunset Review

These regulations will be reviewed in accordance with the sunset review schedule published by the Department to determine whether the regulations effectively fulfill the goals for which they were intended.

H. Regulatory Review

Under section 5(a) of the Regulatory Review Act, the Act of June 30, 1989, (P.L. 73, No. 19), (71 P. S. §§ 745.1-745.15), the Department submitted a copy of the notice of proposed rulemaking, published at Volume 27 Pa. B. 4744 (September 13, 1997) on September 2, 1997 to the Independent Regulatory Review Commission and the Chairpersons of the Senate and House Environmental Resources and Energy Committees for review and comment. In compliance with Section 5(b.1), the Department also provided the Commission and the Committees with copies of all comments received, as well as other documentation.

In preparing this final-form regulation, the Department has considered all comments received from the Commission, the Committees and the public.

This final-form regulation was (deemed) approved by the House Environmental Resources and Energy Committee of	n	and
was (deemed) approved by the Senate Environmental Resources and Energy Committee on	The Commission met on	
and (deemed) approved the regulation in accordance with Section 5(c) of the Act.		

I. Order of the Secretary

Acting under the authorizing statutes, I hereby order that:

- (1) The regulations of the Department of Conservation and Natural Resources, 17 Pennsylvania Code, Chapter 21, are amended by amending Chapter 21 to read as set forth in Annex A, with ellipses referring to the existing text of the regulations.
- (2) The Department shall submit this order and Annex A to the Office of General Counsel and the Office of Attorney General for approval and review as to legality and form, as required by law.

(3)	The Department shall submit this order and Annex A to the Independent Regulatory Review Commission and the Senate and House
	Environmental Resources and Energy Committees as required by the Regulatory Review Act.

- (4) The Department shall certify this order and Annex A and deposit them with the Legislative Reference Bureau as required by law.
- (5) This order shall take effect immediately.

BY:

John C. Oliver Secretary Department of Conservation and Natural Resources

TITLE 17. CONSERVATION AND NATURAL RESOURCES

PART I. DEPARTMENT OF CONSERVATION AND NATURAL RESOURCES

Subpart C. STATE FORESTS

CHAPTER 21. GENERAL PROVISIONS

PRELIMINARY PROVISIONS

§ 21.1. Definitions.

The following words and terms, when used in this chapter, have the following meanings, unless the context clearly indicates otherwise:

[Bureau-The Bureau of Forestry of the Department.]

Camping-Overnight lodging using standard camping equipment.

Commercial activity—An activity in which a person, organization or association directly or indirectly accepts consideration of value as compensation for the provision of goods or services, including transportation.

District Forester-The Department's Bureau of Forestry employe so designated.

Group activity—An organized activity occurring as a scheduled event and requiring a large land area or unique land formation and which may conflict with normal or traditional uses or pose a potential environmental problem.

Pack animal-A horse, mule, donkey or other heavy beast of burden.

State Forest designated hiking trail—Any trail on State Forest land which is listed as a major named hiking trail in the State Forest resource plan and posted as a State Forest designated hiking trail.

State Forest land-Land which is owned or leased by the Commonwealth and which is administered by the Bureau of Forestry.

§ 21.2. Scope.

This chapter applies to State [forests in the Commonwealth under the jurisdiction of the Department] Forest land.

§ 21.3. Violations.

(a) A person is guilty of a summary offense under 18 Pa.C.S. § 7505 (relating to violation of governmental rules regarding traffic) if that person commits an act which is prohibited by § 21.21 (relating to [licensed] motor vehicles).

.

§ 21.4 Closure.

- (a) Areas or portions of a State Forest or State Forest facilities may be closed or restricted to certain uses by the District Forester or a designee.
- (b) The public will be informed of the closed or restricted areas or facilities by any form of communication, which may include posted signs, or by fencing or other enclosure manifestly designed to exclude intruders.
- (c) Violation of the closure or restriction notices under subsection (b) is prohibited.

BOATING

§ 21.11. Use.

[Unless posted otherwise, all] All waters on State Forest land [shall be] are open for the operation of watercraft unless posted as closed.

[Operation of watercraft in waters posted against] [such] [this] [operation] [shall be] [is] [prohibited.]

- § 21.12. Mooring and launching.
- (a) Overnight mooring or storage of watercraft from November 15 through March 31 [shall be] is prohibited except on lands leased from the Commonwealth. This prohibition does not apply to canoes, kavaks, rafts and other paddle-powered small boats stored overnight at the operator's campsite.
- (b) The launching or mooring of watercraft on waters [lakes and ponds] in State Forests requires the appropriate valid State Recreation Area Watercraft Permit affixed aft of amidship on the starboard-right-side.

* * * * *

- (d) [Fees are as follows:
- (1) Mooring Permit \$15/year
- (2) Launching Permit \$5.00/year
- (3) Boat Dock Permit \$5.00/year]

[Small, hand-carried boats, rafts and canoes may be launched from any point on the shoreline. Boats on trailers may be launched only at designated launch areas.] The Department will establish a schedule of fees for mooring and launching permits. This schedule and any subsequent revisions will be effective upon publication in the Pennsylvania Bulletin.

- (e) [Boat launching shall be prohibited at other than designated launching areas.] Small, hand-carried boats, rafts and canoes may be launched from any point on the shoreline. Boats on trailers may be launched only at designated launch areas.
- (f) [(e)] Boat mooring [shall be] is [prohibited at other than] restricted to designated mooring areas.
- § 21.13. Motorized boats.

[Use of motors, except electric motors, for the propulsion of watercraft] [shall be] [is] [prohibited.] Only electric motors may be used for propulsion of motorized watercraft.

MOTORIZED VEHICLES

- § 21.21. [Licensed motor] Motor vehicles.
- (a) This section pertains to motor vehicles other than those addressed in §§ 21.22, 21.23 and 21.23a [and 21.23b] (relating to snowmobiles; [all-terrain vehicles; and all-terrain vehicles).

- (b) [Licensed motor vehicles shall be permitted on State Forest roads open to public travel.] Operation of motor vehicles on State Forest land in the following manners [shall be] is prohibited:
- (1) Operation of [any] a motor vehicle in [a reckless or negligent manner,] careless disregard for the safety of persons or property, or in excess of posted speed limits or, where no speed limit is posted, in excess of 25 miles per hour.
- (2) Driving on roads, trails or other areas not specifically designated and posted for motor vehicle traffic unless authorized in writing by the District Forester or a designee.
 - [(2)][(3)] [Driving on roads, trails] [,] [or] [other] [areas closed to] [vehicular traffic] [motor vehicles] [by posted signs or barriers.]
- [(3)][(4)] The use of State Forest roads for commercial purposes without a Road Use Agreement from the Department. An individual or corporation whose property adjoins State Forest land is not required to obtain a Road Use Agreement to use State Forest roads for incidental commercial purposes to make deliveries to the property.
 - [(4) Parking a vehicle in front of gates or on roads in such a manner as to interfere with the free use of the gate or the road.]
 - [(5)] (4) Operating unlicensed [or], unregistered or uninspected motor vehicles on State Forest [roads open to public travel] land.
- § 21.22. Snowmobiles.
 - (a) Snowmobiling [shall be] is permitted in accordance with 75 Pa.C.S. Chapter 77 (relating to snowmobiles) and this section.
- (b) Snowmobiles may be operated only on designated snowmobile roads, designated trails[,] and designated and posted areas on State Forest land [from the day following the last day of] [regular or extended] [antierless deer season as established by the Game Commission until the following April 1,] [inclusive,] [or earlier, as determined by] [snow conditions] [the District Forester].
- (c) Snowmobiles may be operated on State Forest land from the day following the last day of regular or extended antierless deer season as established by the Game Commission through the following April 1, or earlier, as determined by the District Forester.
- [(c)] (d) Operation of snowmobiles on [any] a road, trail[,] or area not [specially] specifically designated and posted for snowmobile use or at any time of year [in which snowmobiling] [shall] [is] [not] [be] [permitted under this section] [shall be] other than that set forth in subsection (c) of this section is prohibited.
- § 21.23. Trail bikes[,] and other motorized off-road vehicles.

This section pertains to motorized off-road vehicles other than those addressed in §§ 21.22 and 21.23a (relating to snowmobiles; and all-terrain vehicles).

- [(a)] (1) Trail bikes[,] and other motorized off-road vehicles [, and all other similar motorized recreation vehicles] may be operated only on [designated] roads, [designated] trails[,] or other areas [which] that have been specifically designated and posted for their use.
- [(b)] (2) [Unless authorized in writing by the Bureau, all such vehicles are excluded from State Forest land between September 25 and the day following the last day of antierless deer hunting season, inclusive, as established by the Game Commission.] Trail bikes and other motorized off-road vehicles may be operated only during the times of year authorized by the Department's Bureau of Forestry by posted signs.
- [(c)] [(3)] [Operation of trail bikes] [, off-road vehicles,] [and] [all] [other] [similar] [motorized] [recreation] [off-road] [vehicles on any road, trail] [,] [or] [other] [area not] [specially] [specifically] [designated] [and posted] [for their use, or at any time of] [the] [year in which their use is excluded under this section] [shall be] [, is] [prohibited.]

§ 21.23a. All-terrain vehicles.

- (a) The operation of all-terrain vehicles is permitted in accordance with 75 Pa.C.S. Chapter 77 (relating to all-terrain vehicles) and this section.
- (b) All-terrain vehicles may be operated only on designated and posted roads, trails and other areas [from the Friday before Memorial Day until September 24, inclusive, and from the day following the last day of the regular or extended antierless deer season as established by the Game Commission until April 1, inclusive].
- (c) [Operation of all-terrain vehicles on a road, trail or area not specifically designated and posted for all-terrain vehicle use, or at any time of year in which all-terrain vehicle use is excluded under this section is prohibited.] All-terrain vehicles may be operated on State Forest land from the Friday before Memorial Day through the last full weekend in September, and from the day following the last day of the regular or extended antieriess deer season as established by the Game Commission through the following April 1.

§ 21.24. Spark arrestors.

- [(a) All trail bikes, off-road vehicles, and other similar motorized recreation vehicles operating in, on, or through State Forest land shall have an approved, properly installed spark arrestor which shall meet and be qualified to either of the following:
- (1) The United States Department of Agriculture, Forest Service Standard 5100-1a.
- (2) The 80% efficiency level when determined in accordance with the appropriate SAE recommended practices J335 or J350 noise level restrictions.
- (b) The United States Forest Service publication, Spark Arrestor Guide, will be used as the reference sources in determining whether or not a spark arrestor is approved.
- (c) Operation of any trail bike, off-road vehicle, or other similar motorized recreation vehicle in, on, or through State Forest land without an approved spark arrestor shall be prohibited.]

Operation of any motorized vehicle, all-terrain vehicle, trail bike or other motorized off-road vehicle in, on or through State Forest land without a fully functional spark arrestor is prohibited.

§ 21.25. Parking

- (a) Parking a motor vehicle, all-terrain vehicle, snowmobile, trail bike, other motorized off-road vehicle, boat trailer, camp trailer or other equipment which obstructs a gate, road, trail, footpath, bicycle path, access way, drinking fountain, entrance, exit or road turnaround on State Forest land is prohibited.
- (b) No person shall park a vehicle in a parking area on State Forest land designated for handicapped persons, unless the vehicle bears a handicapped or severely disabled veteran registration plate or displays a handicapped or severely disabled veteran parking placard, in accordance with §1338 of the Vehicle Code, and the vehicle is operated by or for the transportation of a handicapped person or a severely disabled veteran.

[(b)](c) Violations of this section will be handled as follows:

- (1) Prior to the filing of a citation charging a summary offense under this section, the Department may issue a parking ticket, which will be handed to the violator or placed on the windshield of the violator's vehicle. The violator may avoid criminal proceedings by paying a charge to the Department, equivalent to the maximum fine as provided in 18 Pa.C.S. § 7505 (relating to violation of governmental rules regarding traffic), within 5 days of the violation in the manner specified on the ticket.
- (2) When a parking ticket has been issued, the Department may institute criminal proceedings only upon failure of the violator to pay the charge in accordance with the time limit specified in paragraph (1) and in the manner specified on the ticket.
 - (3) When a parking ticket has not been issued, the Department may institute criminal proceedings by issuing a citation.

IANIMALS AND NONMOTORIZED VEHICLES

§21.26. Horses and pack animals.

Horses and pack animals are permitted on State Forest land with the following exceptions:

- (1) National and State Forest designated hiking trails unless posted as being open to their use.
- (2) A road, trail or other area posted against the use of the animals.
- § 21.27. Bicycles and person-powered vehicles.

Bicvcles and person-powered vehicles are permitted on State Forest land with the following exceptions:

- (1) National and State Forest designated hiking trails unless posted as being open to their use.
- (2) A road, trail or other area posted against the use of the vehicles. |

[FUEL WOOD]

FOREST PRODUCTS

§ 21.31. [Permit] Prohibitions.

[Cutting and removing any wood from State Forest land without a permit issued by the District Forester or his designee shall be prohibited.] The following activities are prohibited:

- (1) Cutting, picking, digging, damaging or removing, in whole or in part, a living or dead plant, vine, shrub, tree or flower on State Forest land without written authorization of the District Forester or a designee, except that [berries, nuts, leeks and mushrooms] edible wild plants or plant parts may be gathered without authorization if they are gathered for one's own personal or family consumption. Dead and down wood for small campfires may be gathered without prior authorization.
- (2) Removing rocks, shale, sand, clay, soil or other mineral products from State Forest land without written authorization of the District Forester or a designee.
- (3) Removing peat, sawdust, bark, mulch or other products from State Forest land without written authorization of the District Forester or a designee.
- (4) Removing or disturbing historical or archeological resources from State Forest land without written authorization of the District Forester or a designee.
 - (5) Planting a tree, shrub or plant of any kind without the written authorization of the District Forester or a designee.
- § 21.32. Designated trees.

[Cutting or removal of other than designated or marked trees] [shall be] [is][prohibited.] Only designated trees may be cut or removed.

- § 21.33. Cutting practices.
- (a) The stump height of cut trees [shall] may not exceed the diameter of the stump or 12 inches whichever is smaller.

- (b) All tops and slash shall be removed 25 feet from streams, roads, trails and State Forest boundaries.
- (c) [The use of unauthorized cutting practices or the] The blocking of drainage ditches [along forest roads or streams], pipes and other structures with tops and slash [shall be] is prohibited.
- § 21.34. Removal.
- [(a)] The removal of fuel wood and other forest products from the forest by [means] the use of tractors, skidders or any method other than hand carrying [shall be] is prohibited [unless otherwise authorized in writing by] without written authorization of the District Forester or a designee.
 - [(b) All wood of greater than 4-inch diameter from a felled tree must be removed.]
- § 21.35. [Liability] (Reserved).

[The Commonwealth will not be liable for any accidents, injuries or damages incurred by or caused by the permittee in cutting and removing fuel wood.]

MISCELLANEOUS PROVISIONS

§ 21.61. Camping permit.

[Camping is defined as overnight lodging using standard camping equipment.] Camping without a current camping permit issued by the District Forester or [his] a designee is prohibited; provided, however, that primitive backpack campers not using developed facilities shall not need a permit if they stay no more than one night at any campsite.

- § 21.62. Open fires.
- (a) [Small fires for cooking or warming purposes shall be] <u>Subject to the prohibition in subsection (b)</u>, <u>small campfires are permitted only [at places]</u> where adequate precautions are taken to prevent the spread of fire into the forest [; all]. <u>All</u> other fires are prohibited. <u>Campfires shall be attended</u> at all times.
- (b) [All open] Open fires [of any kind shall be] are prohibited when the forest-fire danger is [posted as] determined by the District Forester to be High, Very High or Extreme and from March 1 through May 25 [inclusive] and from October 1 through December 1[, inclusive] without authorization from the District Forester or a designee. This prohibition does not apply to small self-contained camp stoves when used in a safe manner.
- (c) A person causing a wildfire, in addition to possible criminal penalty, [shall be] is liable for all damages, costs of extinction[,] and fines.
- § 21.63. Hunting and trapping.
- [(a)] Hunting [shall be] and trapping on State Forest land are permitted in accordance with [current] the Pennsylvania [game laws] Game and Wildlife Code [on all areas] and regulations, unless otherwise posted.
- [(b) Hunting] [and trapping] [in violation of posted closure or special restriction notices] [shall be] [are] [prohibited.]
- § 21.64. Fishing.
- [(a)] Fishing [shall be] in waters on State Forest land is permitted in accordance with [current] the Pennsylvania [fish] Fish and [boat] Boat [laws] Code and rules and regulations, unless otherwise posted.
 - [(b) Fishing in violation of posted closure or special restriction notices] [shall be] [is] [prohibited.]

§ 21.65. Target shooting.

Target shooting with firearms [or], bows and arrows [at other than protected and approved targets and in conformity with the Pennsylvania game laws shall be prohibited] or devices capable of launching projectiles and causing injury to persons or property is prohibited [at any location where it could cause or threaten injury, damage or disturbance to a person, animal, living tree, sign, structure or equipment] except where authorized by the District Forester or a designee.

§ 21.66. Destruction of property.

Damaging, defacing or removing any sign, structure, equipment or other material [shall be] is prohibited.

§ 21.67. Posting and soliciting.

Posting of signs or soliciting without [the permission of the Department] written authorization of the District Forester or a designee is prohibited.

§ 21.68. Littering and disposal.

Littering [of areas with] or disposal of garbage. paper, household refuse, waste or other [waste] material of any kind is prohibited.

§ 21.69. [Swimming] (Reserved).

[Unless otherwise authorized, swimming shall be prohibited in all dams, ponds, lakes and streams on State Forest lands.]

§ 21.70. [Removal of plants] (Reserved).

[Picking, digging, cutting, damaging or removing any living plant, vine, shrub or tree or flower thereof shall be prohibited unless authorized by written permit from the Department.]

§ 21.70a. Snow plowing.

The plowing or removal of snow from State Forest land without [written] authorization of the District Forester or a designee is prohibited.

§ 21.71. [Closure.] (Reserved.)

- [(a) Areas or portions of a State Forest or State Forest facilities may be closed] [or restricted] [to certain] [specified] [uses by the] [Department] [District Forester or a designee.]
- [(b)] [Restricted areas or facilities will be conspicuously posted to inform the public of the restricted use of the area.] [The public will be informed of the closed or restricted areas or facilities by any form of communication, which may include posted signs, or by fencing or other enclosure manifestly designed to exclude intruders.]
 - [(c) Violation of the closure] [or restriction] [notices under subsection] [(a)] [(b)] [is prohibited.]
- § 21.72. Closure because of fire danger.
- (a) [In the event] If the forest-fire danger rating is Very High or Extreme, [the Director of the Bureau may close] all or portions of the State Forest may be closed to certain specified uses.

* * * * *

§ 21.73. Pets.

Pets are permitted on State Forest land if they are kept under control and attended at all times.

§ 21.74. Disorderly conduct.

The following are prohibited:

- (1) Fighting [,] or other behavior that is threatening, violent or tumultuous [behavior].
- (2) The possession or consumption of alcoholic beverages by persons under 21 years of age.
- (3) Creating an unreasonable noise that may disturb other visitors to State Forest land.

§ 21.75. Sanitation.

[The use of water outlets, springs, lakes or waterways for washing purposes, or the discharge of trailer, camper or motor home sewage, sink water or bath water on or into the ground or surface waters, is prohibited.]

- (a) Washing in water outlets, springs, lakes or waterways is prohibited.
- (b) Discharging of trailer, camper or motor home sewage, sink water or bath water on or into ground or surface waters is prohibited.
- § 21.76. Feeding of wildlife.
- (a) Feeding of wildlife, except for elevated songbird feeders of less than one-half bushel capacity, is prohibited on State Forest land from 30 days prior to the beginning of spring gobbler season [until the day following the last day] through the end of flintlock muzzleloader deer season.
- (b) Placing of wildlife feeders of more than one-half bushel combined capacity on State Forest land without the authorization of the District Forester or a designee, is prohibited.

§ 21.77. Commercial activity.

Selling, distributing, delivering, servicing, guiding or renting any equipment, material or commodity or otherwise engaging in commercial activity on State Forest land without written authorization of the District Forester or a designee is prohibited.

§ 21.78. Group activities.

Group activities are prohibited without written authorization of the District Forester or a designee.

§ 21.79. Structures.

Building or erecting a structure or memorial, or engaging in construction or excavation activity, without written authorization of the Department is prohibited.

§ 21.80. Trespass.

(a) A person who violates this chapter or disregards instructions or warnings given by a State Forest officer or interferes with a State Forest officer in the performance of the duties of the officer may be ordered to leave State Forest land.

(b) A person who refuses to leave State Forest land, after receiving an order to leave from a State Forest officer, commits an act of criminal trespass under subsection (b) of 18 Pa.C.S. § 3503 (relating to [criminal] defiant trespasser).

DEPARTMENT OF CONSERVATION AND NATURAL RESOURCES COMMENT AND RESPONSE DOCUMENT 17 PA CODE CHAPTER 21 STATE FOREST RULES AND REGULATIONS

OCTOBER 1998

Bureau of Forestry
Department of Conservation and
Natural Resources
P.O. Box 8552
Harrisburg, PA 17105-8552

No	Mr/Mrs/Ms	FN.	MI	LILEN	Acdress	City III	State	270
1	Mr.	David	О.	Schiller	3150 East Second St.	Harrisburg	PA	16915-0673
2	Mr.	Craig	S.	Fishel	423 E. Central Avenue	S. Williamsport	PA	17701
3	Mr. & Mrs.			Vronoski	RR 1 117-0	Sigel	PA	15860
4	Mrs.	Laura		Fryzel	P.O. Box 36	Sterling	PA	18463
5	Mr.	Peter	A.	Colangelo	P.O. Box 67000	Harrisburg	PA	17106-7000
6	Mr.	Robert	M.	Behiing	52 Swamp Road	Wernersville	PA	19565
7	Mr.	Floyd	F.	Miller, Jr.	RD 2 Box 500EE	Kunkletown	PA	18058
8	Mr. & Mrs.	James and Anne		Kelhart	5924 Bogart Court	Slatington	PA	18080
9	Mr.	Robert	W.	Brandt, Jr.	RD 1 Box 42-A	Blain	PA	17006
10	The Honorable	J. Doyle		Corman	Senate Box 203034	Harrisburg	PA	17120-3034
11	Mr.	Fred		Maurer	3165 D Street	Philadelphia	PA	19120
12	Mr.	John		Hummel	P.O. Box 207	Kennerdeli	PA	16374
13	Ms.	Susan	M.	May	RR 2 Box 94	Hughesville	PA	17737
14	Ms.	Susan	Y.	Story	RR 2 Box 100	Muncy	PA	17756
15	Mr.	Anthony	D.	Santoli	401 Samters Building	Scranton	PA	18503
16	Ms.	Lavonne		Young	RR 2 Box 81	Williamsport	PA	17701
17	Dr.	George	A.	Durrwachter	48 West Third St.	Williamsport	PA	17701
18	Ms.	Phyllis	M.	Story	RD 2 Box 100	Muncy	PA	17756
19	Mr.	David		Derecola	4250 Doncaster Dr.	Indian Head	MD	20640
20	Ms.	Linda		Flemmer	RR 3 Box 36C	Bruceton Mills	w	26525
21	Ms.	Trudy	S.	Brown	500 Jackson Road	Danville	PA	17821
22	Ms.	Nancy		Summers	15 Pine Tree Drive	Mechanicsburg	PA	17055
23	Mr.	Andrew	C.	Boyden	Box 94 Rt. 287 S	Wellsboro	PA	16901
24	Mr.	Thomas	T.	Thwaites	P.O. Box 167	Boalsburg	PA	16827
25	Ms.	Emily	1	Richardson	1313 North Lynbrook Drive	Arlington	VA	22201
26	Mr.	Anthony	D.	Santoli	401 Samters Building	Scranton	PA	18503
27	Mr.	Haroid		London	HC 31 Box 58A	Williamsport	PA	17701
28	Mr.	Thomas	Α.	Linzey	2244 Lindsay Lot Road	Shippensburg	PA	17257
29	Ms.	Barbara	1	Horstmeier	1233 Reading Avenue	Boyertown	PA	19512
30	Ms.	Ann	R.	Zedaker	5375 Youngstown Poland Rd.	Poland	ОН	44514
31	Ms.	Janet		Derr	220 Pheasant Rd.	Grantville	PA	17028
32	Ms.	Cheryl		Hartman	RR 1 Box 530	Montoursville	PA	17754
33	Mr.	Daniel		Hartman	RR 1 Box 530	Montoursville	PA	17754
34	Ms.	Joan		London	HC 31 Box 58A	Williamsport	PA	17701
35	Mr.	Don	F.	Shannon, Sr		Muncy	PA	17756
36	Mr.	Donald	E.	Shelley	RR 4 Box 389-11	Jersey Shore	PA	17740

No Mr/Mrs/Ms	THE PARTY OF THE P	Mi	HEN	Address (I ME KICKY	State	Zio
37 Mr. & Mrs.	Joseph & Patricia		Merrick	RD 2 Box 313		PA	17758-9342
38 Mr.	James	G.	Golden	RR 1 Box 164	Muncy Valley	PA	17758
39 Mr.	Douglas		Hay	P.O. Box 481	Mill Rift	PA	18340
40 Mr. & Mrs.	Gene & Debbie		Zanot	RD 1 Box 132A	Reynoldsville	PA	15851
41 Mr.	Ronnie		Lukens	817 Meade Street	Williamsport	PA	17701
42 Mr.	James	F.	Connor	HC 1 Box 95A		PA	18370-9723
43 Mr. & Mrs.	David & Landa		Frederick	RR 2 Box 312	Muncy	PA	17756
44 Ms.	Patricia	P.	Russell	HC 31 Box 49C	Williamsport	PA	17701
45 Mr.	Earl	L.	Stroble	RR 4 Box 500	Jersey Shore	PA	17740
46 Mr.	Ronald	L.	Miller	RR 3 Box 152	Williamsport	PA	17701
47 Mr.	Chet		Hemmerte	404 Iroquois Street	Emmaus	PA	18049
48 Ms.	Amy	C.	Fink	RR 1 Box 349A	Linden	PA	17744
49 Ms.	Loretta		Wazelle	5555 Latta Road	East Palestine	ОН	44413
50 Mr.	Michal	C.	Cook	RD 3 Box 100B	Newport	PA	17074
51 Ms.	Joanne		Frantz	46 Main Street	McEwensville	PA	17749
52 Ms.	Mary	N.	Niklaus	3456 Trindle Road	Camp Hill	PA	17011
53 Mr.	Ronald	W.	Niklaus	9 Stillmeadow Lane	Enola	PA	17025
54 Mr. & Mrs.	William & Betty Lo		Ohnmeiss	RD 1 Box 179E	Trout Run	PA	17771
55 Ms.	Lynne		Phillips	1502 E. Third St.	Williamsport	PA	17701
56 Mr.	Clair	L.	Geytgen, Jr.	RD 2 Box 377	Muncy	PA	17756
57 Ms.	Glenda	J.	Taylor	RR 2 Box 296	Muncy	PA	17756
58 Mr.	Edward		Palmatier	RR 7 Box 7280	Muncy	PA	17756
59 Ms.	Jackie	A.	Prentiss	RR 1 Box 274	Linden	PA	17744-9745
60 Ms.	Brenda	H.	Billman	HC 31 Box 319B	Williamsport	PA	17701
61 Mr.	Randy	L.	Billman	HC 31 Box 319B	Williamsport	PA	17701
62 Mr. & Mrs.	Terence & Susan		Schaner	780 Sheep Bridge Rd.	York Haven	PA	17370
63 Ms.	Mary		Trafford	RR 3 Box 114	Williamsport	PA	17701
64 Ms.	Vanessa		Trafford	RR 3 Box 114	Williamsport	PA	17701
65 Ms.	Deb		Bartl	996 Knob Hill Drive	Fayetteville	PA	17222
66 Ms.	Vivian	M.	Huli	RD 2 Box 90	Columbia Cross R	PA	16914
67 Ms.	J. Everly		Smith	RD 3 Box 95	Newport	PA	17074-9656
68 Mr.	George	L.	Gumsey	RR 2 Box 408	Troy	PA	16947
69 Mrs.	Sharon	R.	Gumsey	RR 2 Box 408	Тгоу	PA	16947
70 Ms.	Pamela		Herrington	318 Eldred St.	Williamsport	PA	17701
71 Mr.	J. Glenn		Shawl	RD 2 Box 122	Hughesville	PA	17737
72 Ms.	Emily	D.	Taylor	RD 1 Box 28	Williamsport	PA	17701

No.	Mr/Mrs/Ms		M	LN	LE MILLIA ACCINESE THE REPORT OF THE PERSON	E CICIO	State	Zo III
73 Ms	S .	DeAnna	L.		P.O. Box 2816			17771
74 Ms	S.	David	P.	Baumgardne	RR 2 Box 300-A	Muncy	PA	17756
75 Mr	r.	Charles	N.	Rightmire	RR 1 Box 423		PA	17701
76 Ms	S.	Grace		Miller	RD 1 Box 36		PA	17772
77 Ms	s.	Kimberly		Holohan	RD 1 Box 192			17777
78 Mr	<u></u>	Aaron	S.		3 South Arch St.			17754
79 Mr	Γ.	John	N.	Rathmell	RR 2 Box 345	Muncy	PA	17756
80 Mr	ſ.	David	V.	Cooper	RR 1 Box 143	Montoursville	PA	17754
81 Mr	r.	Joseph	S.	Thompson	1973 Roseville Rd.	Mansfield	PA	16933
82 Mr	<u></u>	Donald	J.	Wells	RR 3 Box 87B	Columbia Cross R	PA	16914
83 Mr	rs.	Barbara	H.	Wells	RR 3 Box 87B	Columbia Cross R	PA	16914
84 Ms		Dianne			RR 1 Box 326			17754
85 Ms		Gillian	 		535 Guardlock Dr.	<u> </u>		17745
86 Ms		Sharon	E.		218 Spring St.			17701
87 Ms		Colleen	M.	Zellers	310 Glenwood Ave.	Williamsport		17701
88 Ms		Betzi	1	Miller	RR 1 Box 144A	Montoursville	PA	17754
89 Ms		Ruth	Ē	Schneider	RR 2 Box 303	Muncy	PA	17756
90 M		Brenda	M.	Garverick	43 Eldred St.	Williamsport	PA	17701
91 M		Betty	S.	Gardner	814 Weldon St.	Montoursville	PA	17754
92 M		Danielle	R.	Maiolo	610 Campbell St.	Williamsport	PA	17701
93 M		John C.	C.	Shuhler	301 E. Homer St.	Michigan City	IN	46360
	r. & Mrs.	Richard & Linda	1	Hanna	1054 Howard Divide Rd.	Howard	PA	16841
95 M		Jeane	1	Parke	RR 1 Box 364	Montoursville	PA	17754
96 M	lr.	Harold	C.	Seyler	302 Buttonwood Ln.	Muncy	PA	17756-8603
97 M	ls.	Sandy	1	Monoski	360 Broad St.	Montoursville	PA	17754
98 M	lr.	Kenneth		Holmes	T.H.V. Box 4059	Montoursville	PA	17754
99 M	is.	Becky	T	Nicklas			PA	
100 M	is.	Michal	M.	Cook	RD 3 Box 100B	Newport	PA	17074
101 M	lr.	Edward	T.	Earley	497 Northway RV Ext.	Williamsport	PA	17701
102 M	ir.	Ezra	J.	Burgess	P.O. Box 403	Picture Rocks	PA	17762
103 M	ls.	Edna		Riddell	RR 1 Box 694	Montgomery	PA	17752
104 M	ls.	Marie		Kutz	6168 Greenbriar Lane	Fayetteville	PA	17222-9678
105 M	ls.	Marilyn		Kilmer	RR 1 Box 121-B	Trout Run	PA	17771
106 M	ls.	Cia		Reis	RR 2 Box 336	Muncy	PA	17756
107 M	lr.	s.		Fenstermac	P.O. Box 1327	Chambersburg	PA	17201
108 M	ls.	Christie	L.	Fortin	414 Yerger Road	Linden	PA	17744

No	Mr/Mrs/Ms	FN	Mi	LN	Address 1		State	Zio 🕯
109		George			400 Beaver Creek Rd.		PA	17019
110	Mr.	Donald		Rosenberger	2600 Possum Hollow Rd.	Coopersburg	PA	18036
111	Ms.	Kelly	L.	Pentz	251 Jordan Ave.		PA	17754
112	Ms.	Barbara		Pentz	251 Jordan Ave.	Montoursville	PA	17754
113	The Honorable	Sandra	J.	Major	House Box 202020	Harrisburg	PA	17120-2020
114	Ms.	Barbara	L.	+	RR 2 Box 280	Huntingdon	PA	16652
115		George	A.	Yams	401 Samters Bldg.		PA	18503
116	Ms.	Virginia		Wolfe	604 E. Emerys Church Rd.	Linden	PA	17744
117	Ms.	Victoria	A.		RR 3	Benton	PA	17814
118	Ms.	Melissa		Michaels	Box 356	Winburne	PA	16879
119	Mr.	John	E.	Wolfe	604 E. Emerys Church Rd.	Linden	PA	17744
120	Ms.	Joan	Y.	Hiller	255 Jordan Ave.	Montoursville	PA	17754
121	Ms.	Lynn	M.	Darlington	RD 1 Box 491	Centre Hall	PA	16828
122	Ms.	Bonnie		Darlington	RD 1 Box 491-A	Centre Hall	PA	16828
123	Ms.	Janet		Langers	210 Lincoln St.		PA	17756
124	Ms.	Cindy	D.	Simcox	100 Hideaway Ln.		PA	17044
125	Ms.	Dee		Callahan	HC 64 Box 217	Trout Run	PA	17771
126	Ms.	Joan		Schneider	RR 2 Box 160	Linden	PA	17744
127	Mr.	Robert	E.	George	R-35 Pottsgrove Rd.	Danville	PA	17821
128	Mr.	Patricia	A.	Beaver	RD 1 Box 268	Allenwood	PA	17810
129	Mr.	Michael	1	Cosgrove	229 Ivy Rock Ln.	Havertown	PA	19083-2827
130	Ms.	Susan	K.	Metka	199 Metka Rd.	Schwenksville	PA	19473
131	Ms.	Angie	1	Hay	38 Houtz Ln.	Port Matilda	PA	16870
132	Mr.	Edward	D.	Rakowsky	421 S. Broad Mtn. Ave.	Frackville	PA	17931
	The Honorable	Allan	ļ	Egolf	House Box 202020	Harrisburg	PA	17120-2020
	Mr.	Lewis	J.	Schmalzle	211 Bridge St.	Narrowsburg	NY	12764-0192
135	Mr.	Thomas	M.	Yerkes	3680 Scotland Rd.	Scotland	PA	17254
136	Ms.	Mary	S.	Kinney	830-C Menno Way	Chambersburg	PA	17201
	Ms.	Jineen	K.	Boyle	163 Northwood Ct.	Lebanon	PA	17042
138	Ms.	Lisa	M.	Long	1856 Holly Dr.	Camp Hill	PA	17011
139	Mr. & Mrs.	Joseph & Betty		Clark	410 S. Washington St.	Mount Vemon	PA	17066
140	Ms.	Bonnie	T	Brown	430 Mt. Zion Rd.	Dillsburg	PA	17019
141	Ms.	Lori		Stewart	886 Pt. Phillips Rd.	Bath	PA	18014
142	Ms.	Sally	T	Leeper	RR 4 Box 405	Mifflintown	PA	17059
	Ms.	Mary	T	Rose-Colley	RR 1 Box 721-A	Williamsport	PA	17701
144	Ms.	Julie	1	Fenstermac	414 S. Market St. #3	Mechanicsburg	PA	17055

No Merkes	rtuske (sE) (se	MI			THE RECEIVE	State	Zlo
145 The Honorable				House Box 202020	Harrisburg		17120-2020
148 Mr.	Ralph		Marks	RD 2	Hughesville		17756
147 Ms.	Ann		Keman	1723 Pentridge Cove	Williamsport	PA	17701
148 Ms.	Frances	B.	McDonald	1545 Grampian Blvd.	Williamsport	PA	17701
149	Blair & Jane	1.	Askey	RR 3 Box 582	Mill Hall	PA	17751
150 Ms.	Bertha	R.	Vargson	RR 1 Box 1900	Gillett	PA	16925
151 Mr.	Max	A.	Reeder	RD 6 Box 162	Muncy	PA	17756
152 Ms.	Patti	A.	Treaster	RD 1 Box 413A	Centre Hall	PA	16828
153 Mr.	Robert	A.	Layman	RD 2 Box 42	Everett	PA	15537
154 Ms.	Evelyn	L.	Zeger	HCR 75 Box 126	McConnellsburg	PA	17233
155 The Honorable	Daniel	F.	Clark	House PO Box 202020	Harrisburg	PA	17120-2020
156 Mr.	John	W.	Stein	P.O. Box 9	Hershey	PA	17033-0009
157 Mr.	Mark		Eckler	20 East Ridge Ave.	Greenville	PA	16125
158 Ms.	Pat			6820 Madison Ave.	W. Mifflin		15122
159Ms.	Judith	E.	Rice	RR 1 Box 185	Montoursville		17754
160 Mr. & Mrs.	Henry & Rowena		Betz	294 Briar Lane	Chambersburg	PA	17201
161 Mr.	Raiph	Н.	Kinter	6312 Chelton Ave.	Harrisburg	PA	17112-2366
162 Mr.	Glenn	L.	Oster	784 Olive St.	Pittsburgh	PA	15237-4882
163 Ms.	Jane		Greber	P.O. Box 61001	Harrisburg	PA	17106-1001
164 Mr. & Mrs.	David & Judy		DeRose	RR 8 Box 320B	Bloomsburg	PA	17815-8331
165 Ms.	Helen		Schick	RR 5 Box 616	Jersey Shore	PA	17740
166 Mr.	Clarence		Fredlund	7901 Chambers Hill Rd.	Harrisburg	PA	17111
167 Mr. & Mrs.	John & Edith		Ryan	5930 Faith Circle	Harrisburg	PA	17112
168 Mr.	William	F.	Thompson	2010 Fox Creek Rd.	Berwyn	PA	19312
169 Mr.	Edward		Kenna			PA	
170Mr.	Hugh	S.	Downing	91 Earlwood Road	Pittsburgh	PA	15235-1902
171 Mr.	Randy	P.	Brungard	379 Marsh Creek Rd.	Howard	PA	16841
172 Ms.	Barbara	<u> </u>	Thwaites	1113 Centre Lane	State College	PA	16801
173 Mr.	Joe		Healey	93 Cedarwood Dr.	Lafiln	PA	18702
174 Ms.	Thyra	C.	Sperry	740 Oak Hill Dr.	Boiling Springs	PA	17007
175 Mr.	Thomas	В.	Murphy	Courthouse Room 406	Williamsport	PA	17701-6570
176 Ms.	Ellen	D.	Fatiuk-Haigh	4540 Emig School Rd.	Thomasville	PA	17364
177 Mr.	Frank		Aitken	1813 E. Ontario St.	Philadelphia	PA	19134
178 Ms.	Sara	Y.	Squier	608 E. Center St.	Ulysses	PA	16948
179 Mr.	Wil		Ahn	P.O. Box 643	Coudersport	PA	16915
180 Ms.	Marie		Cass	P.O. Box 304	Ulysses	PA	16948-0304

No	Mr/Mrs/Ms		Mi				State	E Zio
181		Robert	H.		P.O. Box 311	Ulysses	PA	16948-0311
182	Mr.	Allen	J.	Cole	1 Purple St.	Wellsboro	PA	16901
183	Mr.	Skee		Derr	424 Blacklatch Lane	Camp Hill	PA	17011-8414
184	Mr.	Lamar		Powell	6728 Furrow Dr.	Harrison	TN	37341
185	Mr	Richard	-	Menoche	321B Long Highway	Little Compton	\- 	02837
186		Edward	В.		4010 Crayton Rd.	Naples	FL	34103-3533
187		Carolyn			111 Estella Dr.	Reedsville		17084
188		E. Scott		·	RD 1 Box 369	Centre Hall		16828
189		Jana	<u> </u>		411 W. Gatesburg Rd.	Penna. Furnace		16865
190		David			P.O. Box 24	Warriors Mark		16877
191		Jay			630 South Linden	Turbotville	PA	17772
192		Bonnie	M.		253 Rockdale Rd.	Butler	PA	16002
193		Kurt	 		RD 1 Box 399	Alexandria	PA	16611-9411
194		Rhonda	 	Cole	P.O. Box 511	Centre Hall	PA	16828
195		Cindy	 		RD 2 Box 70	Spring Mills	PA	16875
196		John	M.		Box 385	Turbotville	PA	16828
197		Alfred	S.		RD 2 Box 209	Turbotville	PA	16828
198	Ms.	Jennifer	1	Bielesch	8632 Yale Place	Philadelphia	PA	19136
199	Mr.	Richard	E.	Bland	435 E. Irvin Ave.	State College	PA	16801
200	Ms.	Kimberly	K.	DeRosia	473 Baker Rd.	Freedom	PA	15042
201	Ms.	Luann		Bamhart	RD 2 Box 342-A	Blairsville	PA	15717
202	Ms.	Jill		Haugh	119 Oak Hill Rd.	West Sunbury	PA	16061
203	Mr.	Harold	G.	Eyster	190 Kings Highway	Altoona	PA	16602
204	Ms.	Christine	L.	McCahren	RR 1 Box 205	Port Royal	PA	17082
205	Mr.	Robert	E.	Mikesell	RR 2 Box 2095	Port Royal	PA	17082
206	Ms.	Dawn	1	Reber	P.O. Box 295	Minersville	PA	17954
207	Ms.	Naomi		Younger	204 Stonehedge Rd.	Hollidaysburg	PA	16648-9777
208	Mr.	J. William		Conway	1306 Linn St.	State College	PA	16803
209	Ms.	Deborah	J.	Lakis	2448 W. Mountain Ave.	S. Williamsport	PA	17701
210	Ms.	Sharon	В.	Albertson	3383 Harvest Rd.	Elizabethtown	PA	17022
211	Ms.	Frances	R.	Haube	830 Blanchard St.	Bellefonte	PA	16823
	Ms.	Henrietta	A.	Palmer	55 Bittle Rd.	Littlestown	PA	17340
	Ms.	Linda	M.	Myers	RD 1 Box 198	Petersburg	PA	16669
214		George	A.	Luse, Jr.	RD 2 Box 14	Spring Mills	PA	16875
	Ms.	Cheryl		Zimmerman	854 Washington Ave.	Northampton	PA	18007
216	Ms.	Jeanne	T.	Six	202 Cartwright Rd.	Chicora	PA	16025

No Mr/Mrs/Ms		M	UN	Acdress 1	E GREW	State	Zip
217 Ms.	Helen	E.		Box 465	Beech Creek	PA	19822
218 Mr.	Grant	W.	Sherritt	131 E. Lyttle Ave.	State College	PA	16801
219 Ms.	Betty	M.	Kraus	590 Saxonburg Rd.	Butler	PA	16002-3710
220 Ms.	Donna		Rovenolt	RD 2 Box 209	Turbotville	PA	17772
221 Mr.	James	W.	Wright	RD 6 Box 433	Muncy	PA	17756
222 Ms.	Kathryn	J.		RR 1 Box 145A	Trout Run	PA	17771
223 Ms.	Victoria		Swayne	1380 Mountain Rd.	Manheim	PA	17545
224 Ms.	Deborah		Walter	RR 3 Box 61	Troy	PA	16947
225 Ms.	Barbara	J.	Hood	HCR 62 Box 127	Renovo	PA	17764
226 Ms.	Karen	M.	Lamper	RD 1 Box 232	Muncy		17756
227 Ms.	Wendy	R.		RR 2 Box 29	Linden		17744
228 Ms.	Lynda	L.	Phillips	921 Washington Blvd.	Williamsport	PA	17701
229 Ms.	Irene	C.	Harrison	92 Carone Dr.	Cogan Station	PA	17728
230 Mr.	Jeffery	A.	Easter	715 Campbell St.	Williamsport	PA	17701
231 Mr.	D.		Lingertot	RD 2 Box 138C	Cogan Station	PA	17728
232 Ms.	Ellen		Bran	480 Rt. 793 W.	Cogan Station	PA	17728
233 Ms.	Betty	J.	Bubb	HC 31 Box 3200	Williamsport	PA	17701-9790
234 Ms.	Patricia		Sydak	427 Hawthome Ave.	Williamsport	PA	17701-9790
235 Ms.	Janet		Bamhart	110 Wilson St.	Jersey Shore	PA	17701
236 Mr.	David	W.	Early	458 Birch St.	Lock Haven	PA	17745
237 Ms.	Dee	A.	Coffey	373 Lyons Ave.	Williamsport	PA	17701
238 Ms.	Robin			RR 2 Box 520	Montgomery	PA	17752
239 Ms.	Susan		Alpaugh	RD 5 Box 232	Muncy	PA	17756
240 Ms.	Shelly	J.	Kriner	2037 Sechler Dr.	Williamsport	PA	17701
241 Ms.	Mary	A.	Shaner	RR 4 Box 182	Muncy	PA	17756
242 Mr.	Christopher		Tobias	RD 3 Box 682	Williamsport	PA	17701
243 Ms.	Diane	L.	Miller	RR 7 Box 175	Muncy	PA	17756
244 Ms.	Mandy		Middaugh	RR 5 Box 314	Muncy	PA	17756
245 Ms.	Clarissa		Miller	HC 31 Box 191A	Williamsport	PA	17701
246 Ms.	Rebecca		Laidacker	RD 1 Box 235C	Watsontown	PA	17777
247 Ms.	Colette	L.	Richards	714 Seminary St.	Jersey Shore	PA	17740
248 Ms.	Maxine	E.	Wingrove	110 Dartmouth St.	Williamsport	PA	17701
249 Mr.	Richard	W.	Langer	210 Lincoln St.	Muncy	PA	17756
250 Ms.	Jean			211 Sherman St.	Muncy	PA	17756
251 Ms.	Vivian		Mills	RR 1 Box 437	Hughesville	PA	17737
252 Ms.	Terri	Α.	Shick	Box 4033 Lot B-31	Montoursville	PA	17701

No	Mr/Mrs/Ms		M	HENE	· · · · · · · · · · · · · · · · · · ·			Zio I
253	Ms.	Renee		Snyder	RD 2 Box 200	Montgomery		17752
254	Ms.	Char		Somits	304 Sherman St.	Muncy	PA	17756
255	Ms.	Kathy	J.	Smith	21 S. Main St.	Muncy	PA	17756
256	Mr.	John		Terry	RR 4 Box 141	Jersey Shore	PA	17740
257	Mr.	Fred	L.	Hartman	RR 2 Box 30	Linden	PA	17744
258	Mr.	Larry	K.	Moon	431 W. 8th. Ave.	So. Williamsport		17701
259		Karen		Wright	RR 1 Box 186	Montoursville		17754
260	Ms.	Pamela		Steppe	RR 3 Box 183	Williamsport		17701
261	Mr.	Terry	+	Krezmer	RR 4 Box 276	Montoursville		17754
262		Raymond	S.	Haines	RR 2 Box 504	Jersey Shore	4	17740
263	Mr.	Timothy	E.	Eck	RR 1 Box 268-A	Hughesville	PA	17737
264		Mary	Α.	Smeltzer	RR 1 Box 355-A	Linden	PA	17744
265		Stephen			P.O. Box 254	Woolrich		17779
266		Scott		Boring	RR 2 Box 617	Jersey Shore		17740
267		Gary	L .	Rothmell	1012 Eckard Rd.	Cogan Station		17728
268		Eugene		Hennigan	1681 Taylor Pl.	Williamsport		17701
269		LaVern	<u> </u>	Steppe	RR 3 Box 183	Williamsport	PA	17701
270		Richard	D.	Martin	144 1/2 Memorial Ave.	Williamsport	PA	17701
271		Lori		Tobin	106 Lawson St.	Williamsport	PA	17701
272	Mr.	Thomas	J.	Kuhns	1841 Nast Ave.	Montoursville	PA	17754
273	Ms.	Michele	A.	VanBuskirk	RR 6 Box 85	Muncy	PA	17756
274	Ms.	Linda	L	Rhodes	RR 5 Box 6	Williamsport	PA	17701
275	Mr.	B. Karl		Saydlowski	936 Hollywood Circle	Williamsport	PA	17701
276	Ms.	Esther		Hand			PA	
277	Mr.	Roger	L.	Buck	105 Meadow St. Lot 37	Mansfield	PA	16933
	Mr. & Mrs.	Richard & Helen		Wells	261 Township Line Rd.	Schwenksville	PA	19473
	Mr.	Steve		Roberts	Dept. 2040-1950	Slippery Rock	PA	16057-1326
280	Mr. & Mrs.	Francis & Karen	<u> </u>	DeNaro	RR 1 Box 133-A	Elliottsburg	PA	17024
281	Mr.	Chris		Vreeland	6315 High St.	Haslett	MI	48843
282	Mr.	James	C.	Austin	297 Rhodes Way	Luray	VA	22835
283	Mr. & Mrs.	& Mrs. Donald		Топеу	HC 01Box 69	Shinglehouse	PA	16748
	Mr.	Joseph	W.	Larue	1107 Baldwin St.	Williamsport	PA	17701
	Ms.	Susan	C.	Montler	1796 Bellemeade Dr.	Altoona	PA	16602-7450
	Mr.	Carmen	<u>T.</u>	Santasania	1410 Charles St.	State College	PA	16801
	Ms.	Joan		Duffield	108 Crider Ave.	Fayetteville	PA	17222
288	Mr.	Dennis	E.	Mann	P.O. 147	Roxbury	PA	17251

No	Mr/Mrs/Ms		MI	EN	Accorase (E CRY	State	Zip
289 M		Susanne			5801 Mayfair dr.	Harrisburg		17112
290 M	r.	Curt		Weinhold	704 Southeast St.	Coudersport	PA	16915-1420
291 M	r.	Gregory	W.	Grove	RD 1 Box 483	Petersburg	PA	16669
292 MI	ſ.	Edward		Sohl	802 W. 11th. St.	Wilmington	DE	19801
293 M		Charlotte	lī.	Freeman	P.O. Box 243	Ulysses	PA	16948-0243
294 MI		Jonathan		Kohn	415 Hopl Dr.	Mechanicsburg		17055
295 Ms	3 .	Dorothy	B.	Erway	314 Port Allegany Rd.	Coudersport	PA	16915
296 Mr	r.	William	R.	Menke	700 Rayovac Dr.	Madison	WI	53711
297 Ms	S.	Rita		Bowman	1090 K-mille Rd.	Stevens	PA	17578
298 Ms	S .	M. Virginia		Musser	130 N. 2nd. St. #A-7	Souderton	PA	18964
299 Mr	۲.	Dan		Wolfe	P.O. Box 501	Williamsport	PA	17701
300 Mi	г.	David	N.	Barr	43 Walnut Trail	Carroll Valley	PA	17320
301 M	r.	Mike		Sausser	1608 Long Run Rd.	Schuylkill Haven	PA	17972
302 M	r.	George	G.	Probst	20 Park Ln.	Wellsville	NY	14895
303 M	s.	Susan	J.	Gurekovich	RD 2 Box 28-A	Spring Mills	PA	16875
304 M	r.	David	A.	Lambert	RR 2 Box 504	Muncy	PA	17756
305 Mi	r.	George	L.	Bower	17 New St.	Muncy	PA	17756
306 M	ls.	Grace	1	Yearick	RR 3 Box 559	Mill Hall	PA	17751
307 M	r.	F. Daniel	1	Fry	1184 Russell Ave.	Williamsport	PA	17701
308 M	ls.	Cynthia	M.	Lyons	RR 1 Box 112	Unityville	PA	17774
309 M		Janet	В.	Benner	360 N. Fillmore Rd.	Bellefonte	PA	16823
310 M		Fred	R.	DeFiore	139 Angel Rd.	New Wilmington	PA	16142
311 M	ir.	John	R.	Reed	841 Arlington Ave.	New Castle	PA	16101
312 M	lr. & Mrs.	Gene	1	Zanot	RD 1 Box 132A	Reynoldsville	PA	15851
313 M	lr.	Byron	E.	Shrader	137 McKean Rd.	Freeport	PA	16229
314 M		Willam	J.	Barker	RD 2 Box 246B	Kittanning	PA	16201
315 M	lr.	George	C.	Kauffman, J	197 Second St.	Belleville	PA	17004
316 M	lr.	Joseph	R.	Hood, Jr.	HCR 62 Box 127	Renovo	PA	17764
317 M	ls.	Geraldine		Saunders	RR 1 Box 253A	Canton	PA	17724
318 M	ls.	Patricia		Baker	2115 Blair St.	Williamsport	PA	17701
319 M	lr. & Mrs.	Steve & Jeanne		Askey	RR 1 Box 231	Beech Creek	PA	16822-9507
320 M	ls.	Kay		James	Box 108	Mohrsville	PA	19541-0108
321 M	lr.	Jack	L.	Acker	849 Midway Rd.	Bulger	PA	15019
322 M	ls.	Donna	1	Schrader	137 McKean Rd.	Freeport	PA	16229
323 M		Mary	C.	Schoenberg	761 James Hill Rd.	Columbia	PA	17512
324 M		Kristine	Α.	Dix	RR 1 Box 525E	Centre Hall	PA	16828

Ne	Ne Mis Mis	HAMES NEW TOTAL	M		HTT 11 11 11 11 11 11 11 11 11 11 11 11 1		State	Zio
	Ms.	Catherine	J.	Lowe	P.O. Box 123	Cogan Station	PA	17728
326	Mr.	Steve		Hower	RR 4 Box 335A	Jersey Shore	PA	17740
327	Mr.	Gordon		Taylor	RR 1Box 28	Williamsport	PA	17701
328	Ms.	Linda		Frey-Bulett	404 S. Plum St.	Mount Joy	PA	17552
329	Mr.	ira	J.	Landis	1953 Camp Rd.	Manheim	PA	17545
330	Mr.	Kevin	W.	Kieffer	721 N. Shippen St. 2FR #4	Lancaster	PA	17602
331	Ms.	Michele	L.	Burns	1 Feigles Rd.	Muncy	PA	17758-1708
332	Ms.	Dorothy	L.	Myers	886 E. College Ave.	Bellefonte	PA	16823
333	Mr.	Albert	F.	Newberry, Jr	252 Bennage Ave.	Lock Haven	PA	17745
334	Ms.	Rosalind	K.		405 W. Church	Lock Haven	PA	17745
335	Ms.	Rolinda	F.	Hanley	RR 2 Box 362	Mill Hall	PA	17751
336	Ms.	Teresa	A.	VanGorder	RD 1 Box 103	Mill Hall	PA	17751
337	Mr.	Lon	C.	Pyers	RR 2 Box 136	Lewisburg	PA	17837
338	Ms.	Yvonne	-	Riggs	26 N. Perry St.	Elizabethtown	PA	17022
339		Mitzi			306 Upper Mulberry St.	Danville	PA	17821
340	Ms.	Stacy	Z.	Foulsham	P.O. Box 117	Lock Haven	PA	17745
341	Mr.	Edward	J.	Cook	8 Birch Ln.	Massapequa	NY	11762
342		Barbara	P.	Cook	8 Birch Ln.	Massapequa	NY	11762
343	Ms.	Jeri	L.	Redding	370 New St.	Pottstown	PA	19464
344	Mr.	Lynwood		Redding	370 New St.	Pottstown	PA	19464
345	Ms.	Debbie		Myers	RD Box 1628	Mohnton	PA	19540
348	Ms.	Sherry		Waltman	RD 1 Box 1557	Mohnton	PA	19540
347	Mr.	Barry		Kieffer	675 Sportsman Rd.	Denver	PA	17517
348	Ms.	Margaret	A.	Kieffer	675 Sportsman Rd.	Denver	PA	17517
349	Ms.	Jane		Barenborg	515 Grings Hill Rd.	Sinking Springs	PA	19608
350	Mr.	Melvin		Egolf	1753 Acom Dr.	Sinking Springs	PA	19608
351	Ms.	Andrea	R.	Carter	92 Chapel Hill Rd.	Sinking Springs	PA	19608
352	Ms.	Mary	A.	Brown	Box 2125	Sinking Springs	PA	19608
353	Ms.	Colleen		Krause	102 W. Main St.	Newmanstown	PA	17073
354	Mr.	Jeffrey	C.	Krause	102 W. Main St.	Newmanstown	PA	17073
355	Ms.	Christine		Reber	36 Second St.	Shillington	PA	19607
356	Ms.	Janet	A.	Yeich	129 Franklin St.	Shillington	PA	19607
357	Мг.	Michael	J.	Amoroso	131 Philadelphia Ave.	Shillington	PA	19607
358	Ms.	Kathy		Brown	890 E. College Ave.	Bellefonte	PA	16823
359	Mr.	Gary	1	Brown	890 E. College Ave.	Bellefonte	PA	16823
360	Mr.	Jack	H.	Wertz	823 Main St.	Shoemakersville	PA	19555

No Mr/Mrs/Ms		M		HE WAS IN SECTION AND A SECOND SECTION AND A SECOND SECTION AS A SECOND SECOND SECTION AS A SECOND		State	20
361 Ms.	Dorothy		Wertz	823 Main St.	Shoemakersville	PA	19555
362 Ms.	Mary		Hepler	807 Harmony Hill Rd.	West Chester	PA	19380
363 Ms.	Lee	L.	Hepler	807 Harmony Hill Rd.	West Chester	PA	19380
364 Mr.	Arthur	w.	Shannon	1156 Marshallton Thorndale Rd.		I	19380
365Ms.	Melinda		Shannon	1156 Marshallton Thorndale Rd.			19380
366Mr.	Christopher		Earhart	 	Ephrata		17522
367 Mr.	James		McGovern		Ephrata		17522
368 Mr.	Gregory			4629 Redwood Ln.	Nazareth		18084
369 Ms.	Rachel		Schulze	4629 Redwood Ln.	Nazareth	PA	18064
370 Ms.	Mary Jo			4629 Redwood Ln.	Nazareth		18064
371 Ms.	Sharon		Wagner	525 Berks Pl.	W. Lawn		19609
372 Ms.	Carol		Chabinec	2402 Lindale	W. Lawn		19609
373 Ms.	Angle	J.	Zeller	24 Smith Lane	Bemville	PA	19506
374 Ms.	Donna		Unger	635 Grange Rd.	Bernville	PA	19506
375 Mr.	Craig	 	Jung	P.O. Box 457	Bemville	PA	19506
376 Ms.	Karen	H.	Howeli	58 Bitting Rd.	Alburtis	PA	18011
377 Mr.	Robert	V.	Howeli	58 Bitting Rd.	Alburtis	PA	18011
378 Ms.	Kay	 	Quier	7546 Dresher Rd.	Kempton	PA	19529
379 Ms.	Chris		Quier	7546 Dresher Rd.	Kempton	PA	19529
380 Mr.	Jason	†	Hames	110 St. Andrews Circle	Reading	PA	19607
381 Ms.	Sarah	G.	Moyer	127 Bernhart Ave.	Reading	PA	19605
382 Mr.	Raymond	C.	Trievel	721 Schoffer Rd.	Reading	PA	19606
383 Ms.	Marian		Trievel	721 Schoffer Rd.	Reading	PA	19606
384 Ms.	Lorri	 	Chorgzy	1342 N. 14th. St. Apt. S-23	White Hall	PA	18052
385 Mr.	Steven		Quide	1342 N. 14th. St. Apt. S-23	White Hall	PA	18052
386 Mr.	Lee		Miller	329 White Oak Ln.	Leesport	PA	19533
387 Mr.	Howard		Miller	329 White Oak Ln.	Leesport	PA	19533
388 Ms.	Linda	L.	Evans	731 W. Main St.	Leesport	PA	19533
389 Ms.	Josephine		James	856 Center Rd.	Leesport	PA	19533
390 Ms.	Bonnie		Fulk	827 N. 8th. Ave.	Lebanon	PA	17046
391 Mr.	Stuart		Hanford	3364 W. Oak St.	Lebanon	PA	17042
392 Mr.	Kevin	S.	Moyer	RD 1 Box 141	Robesonia	PA	19551
393 Ms.	Dee	<u> </u>	Moyer	RD 1 Box 141	Robesonia	PA	19551
394 Ms.	Sherry	E.	Radka	RD 1 Box 141	Robesonia	PA	19551
395 Mr.	Lamy	E.	Fultz	1303 Milestone Rd.	Robesonia	PA	19551
396 Ms.	Veda		Lipshaw	33 Ketner's Rd.	Pottsville	PA	17901

Nö	Mr/Mrs/Mr		Mi	THEN.		上海。113 次 距離開	State	W 20 E
397	Ms.	Jennifer			305 Park Rd.	Valley View		17983 ·
398	Mr.	Frederick	W.	Borgmann	Box 302	Geigertown	PA	19523
399	Ms.	Carolyn		Cook	939 Walnut Dr.		PA	18088
400	Ms.	Mollie	B.	Ruth	591 E. Jackson St.	New Holland	PA	17557
401	Ms.	Barbara		Liebhaber	1473 Wethersfield Dr.	Allentown	PA	18104
402	Mr.	Jason		Keeney	RD 2 Box 147A			18833
403		Erin						19530
404		Judith		Kramlich		Limekiln		19535
405	Mr.	Bob		Snyder	100 Elmwood Blvd.			17403
406	Mr.	Jason		King	1116 Deer Run	Orefield	PA	18069
407	Ms.	Mitzi	L.	Mast	334 Snyder Rd.	Oley	PA	19547
408	Mr.	Jeffrey		Parsons	2 Koby Court	Birdsboro	PA	19508
409	Ms.	Kathleen	S.	Lawler	941 Park View Dr.	King of Prussia	PA	19406
410	Ms.	Sheree		Roth	1903 Creek Rd.	_ 	PA	17545
411	Ms.	Mary	A.	Emerich	5107 Mohave Rd.	Temple	PA	19560
412	Mr.	Raymond		Trievel, Jr.	4505 Fox Pointe Ct.	Glen Mills	PA	19342
413		Ralph	R.	Moest	853 Fairmont Ave.	Wemersville	PA	19565
414	Ms.	Jennifer		Elms	253 Race St.	Macungie	PA	18052
415	Mr.	Bill		Brenner	1975 Krammes Rd.	Quakertown	PA	18951
416	Mr.	Gregory	В.	Collen	203 Fawn Ct.	Hamburg	PA	19526
417	Ms.	Tonya		Hendrick	P.O. Box 527	Leesport	PA	19533
418		Roger		lmes, jr.	425 Walnuttown Rd.	Fleetwood	PA	19522
419		Nancy	J.	Ulrich	44 Walnut St.	Adamstown	PA	19501
420	Ms.	Chantelle	M.	Stankiewicz	3304 Harrison Ave.	Riverview Park	PA	19605
421	Ms.	Michelle	L.	Emery	101 Bridge St.	Birdsboro	PA	19508
422	Ms.	Susan	C.	Mont	1796 Bellemeade Dr.	Altoona	PA	16602
423	The Honorable	John	M.	Perzel	110 Main Capitol Bldg.	Harrisburg	PA	17120-2020
424	Mr.	Calvin	W.	DuBrock	2001 Elmerton Ave.	Harrisburg	PA	17110-9797
425	Mr.	J. William		Conway	1306 Linn St.	State College	PA	16803
426	Mr.	George	A.	Luse, Jr.	RD 2 Box 414	Spring Mills	PA	16875
427	Mr.	Stanley	E.	Helt	2005 Reach Rd.	Williamsport	PA	17701
428	Ms.	Debra	K.	Pollack	HC 31 Box 50A	Williamsport	PA	17701
429	Ms.	Mary	L.	Doebler	RR 180x 27	Williamsport	PA	17701
430		Margie	H.	Doebler	RR 1Box 29	Williamsport	PA	17701
431	Ms.	Lisa		Hnatin	935 Louisa St.	Williamsport	PA	17701
432	Dr.	Steven		Winton	1828 E. Third St.	Williamsport	PA	17701

No il il Mindre Ma	群連 建二基位 建建油			開開開開工会議員		340	MIZO
433 Ms.	Patricia		Girven	RR 4 Box 220-A	Towanda	PA	18848
434 Mr.	James	E.	Eicher, Jr.	106 Spruce St.	Towanda	PA	18848
435Ms.	Elizabeth	T	Elcher	106 Spruce St.	Towanda	PA	18848
436 Mr.	Jerry	C.	Sandel	RD 9 Box 364	Bloomsburg	PA	17815-9172
437 Ms.	Joellyn	S.	Sandel	RD 9 Box 364	Bloomsburg	PA	17815-9172
438Ms.	Rita	J.	Spatz	91 Shartlesville Rd.	Mohrsville	PA	19541
439Ms.	Dolores	E.	1	91 Shartlesville Rd.	Mohrsville	PA	19541
440Mr.	Harold		Spatz	552 Irish Creek Rd.	Mohrsville	PA	19541
441 Ms.	Barbara	A.	Spatz	552 Irish Creek Rd.	Mohrsville	PA	19541
442 Ms.	Susan	Н.	Knipe	130 Yorkshire Rd.	Palmyra	PA	17078
443 Mr.	Kenneth	D.	Knipe	130 Yorkshire Rd.	Palmyra	PA	17078
444 Ms.	Linda	N.		RD 1 Box 297-B	Sunbury	PA	17801
445 Ms.	Lisa		Derrick	RD 3 Box 110B	Sunbury	PA	17801
446 Ms.	Cathie		Graham	P.O. Box 12	Blanchard	PA	16826
447 Ms.	Kristie		Graham	P.O. Box 12	Blanchard	PA	16826
448 Ms.	Jennifer		Hoffman	RD 4 Box 925	Mill Hall	PA	17751-9742
449 Mr.	John	D.	Murray	RR 2 Box 804	Hughesville	PA	17737
450Mr.	John	E.	Golder	RR 2 Box 207	Hughesville	PA	17737
451 Mr.	Edward	Ā.	Charles	361 S. 2nd. St.	Hughesville	PA	17737
452 Mr. & Mrs.	John & Linda		Weaver	406 A Valley View Rd.	Myerstown	PA	17087
453Ms.	Anna	R.	Shaak	3 Quarryland Dr.	Myerstown	PA	17067
454 Ms.	Charlotte	V.	Lake	P.O. Box 662	McConnellsburg	PA	17233
455 Mr.	William	E.	Lake	P.O. Box 662	McConnellsburg	PA	17233
456 Ms.	Marsha		Mantzoros	408 W. Church St.	Lock Haven	PA	17745
457 Mr.	Jeffrey	A.	Snyder	11 Cardinal Dr. E.	Lock Haven	PA	17745
458 Ms.	Joy		Fox	RD 1 Box 131AA	Mill Hall	PA	17751
459 Mr.	Ralph	G.	Fox	RD 1 Box 131AA	Mill Hall	PA	17751
460 Mr.	Brad	A.	Fox	RD 1 Box 131AA	Mill Hall	PA	17751
461 Ms.	Jody	L.	Fox	RD 1 Box 131AA	Mill Hall	PA	17751
462 Mr.	Brian	L.	Klopp	415 Rocky Ridge Rd.	Denver	PA	17751
463 Mr.	Eric	S.	Weidenham		Reading	PA	19604
464 Ms.	Margaret	F.	Kauffman	460 Sandy Hill Rd.	Denver	PA	17517
465 Mr.	James	M.	Kauffman	460 Sandy Hill Rd.	Denver	PA	17517
466 Ms.	Kimberly	L.	Firestone	393 Reinholds Rd.	Denver	PA	17517
467 Mr.	Lawrence	Α.	Smith	615 Horseshoe Trail Rd.	Denver	PA	17517
468 Ms.	Nancy	L.	Powell	HC 3 Box 99	Marlenville	PA	16239

No Minnish		M				State	原设
469 Mr.	Donald	R.	Powell	HC 3 Box 99	Marienville	PA	16239
470 Mr.	David	E.	Mechtly	RR 4 Box 186	Montoursville	PA	17754
471 Mr.	Thomas	E.	Garber	RR 1 Box 224	Montoursville	PA	17754
472 Ms.	Pat		Wiseman	206 N. Loyalsock Ave.	Montoursville	PA	17754
473 Ms.	Vickle	Ε.	Garber	RR 1 Box 224	Montoursville	PA	17754
474 Mr.	Richard	G.	Grugan	2028 Warrensville Rd.	Montoursville	PA	17754
475 Mr.	Brent			485 Bernheisel Bridge Rd.	Carlisle		17013
476 Ms.	Nancy			485 Bemheisel Bridge Rd.	Carlisie		17013
477 Mr.	Samuel	C.	Miller	51 Church Rd.	Airville	PA	17302
478 Mr.	Stephen	J.	Krick	11721 Muddy Creek Rd.	Airville	PA	17302
479 Ms.	Dolores		Krick	11721 Muddy Creek Rd.	Airville	PA	17302
480 Mr.	Stephen	J.	Krick	11721 Muddy Creek Rd.	Airville	PA	17302
481 Ms.	Gwen		Wills	RD 2 Box 300	Summerville	PA	15884
482 Mr.	Brian	J.	Bennett	RR 5 Box 315	Muncy	PA	17756
483 Mr.	John	T.	Gramling	RD 5 Box 50	Muncy	PA	17756
484 Mr.	David	М.	Burgett	RD 4 Box 124-3	Muncy	PA	17756
485 Mr.	Frayer		Page	228 Pepper St.	Muncy	PA	17756
486 Mr.	Peter	A.		RD 5 Box 179	Muncy	PA	17756
487 Ms.	Katie		Gresh	RR 2 Box 312	Muncy	PA	17756
488 Ms.	Jiillan		Hafer	RD 1Box 297B	Sunbury	PA	17801
489 Mr.	Scott	R.	Derrick	RD 3 Box 110B	Sunbury	PA	17801
490 Mr.	William	F.	Beaton	P.O. Box 735	Bellefonte	PA	16823
491 Mr.	Robert			209 Governor Park Rd.	Beliefonte	PA	16823
492 Mr.	David	B.	Mackenzie	359 Pleasant Dr. Apt. B	Aliquippa	PA	15001
493 Mr.	Arthur	M.	Wilson	189 Hackney Station Rd.	Amity	PA	15311
494 Ms.	Debra	J.	Mitchell	RR 1 Box 304	Beech Creek	PA	16822
495 Mr.	Richard		Koehler	210 W. Broad St.	Bethlehem	PA	18018
496 Ms.	Debra		Wallace-Har	195 Water St.	Boyertown	PA	19512
497 Ms.	Paula	L.	Malobicky	943 Morgan St.	Brackenridge	PA	15014
498 Ms.	Margaret	T.	Heinterlang	119 Blakely Rd.	Butler	PA	16002
499 Mr.	Barry	L.	Saunder	RR 1 Box 253A	Canton	PA	17724
500 Ms.	Patricia	A.	Neff	RR 3 Box 417A	Cogan Station	PA	17728
501 Ms.	Cathy	A.	McElhinny	RR 1 Box 5E	Corsica	PA	15829
502 Ms.	Florence	Α.	Clever	RD 2 Box 138	Dayton	PA	16222
503 Mr.	Ronald	L.	Wenger	24 Gardner Rd.	Duncannon	PA	17020
504 Mses.	Jo Ann & Gabrielle		Russo	RR 5 Box 5440	East Stroudsburg	PA	18301-9258

No Marinira/Ma		M				State	Zb
505 Ms.	Sarita	L.	Weimer	46 Chestnut Run	Elizabethtown	PA	17022
506 Mr.	William	L.	Shue, Jr.	138 Marshall St.	Ephrata	PA	17522
507 Mr.	Daniel	D.	Schrum	RD 2 Box 4120	Felton	PA	17322
508 Mr. & Mrs.	Dale & Kathy	1	Martin	172 Spook Ln.	Fleetwood	PA	19522
509Ms.	Margo		Hoffman	201 County Line Rd.	Fredericksburg	PA	17026
510Ms.	Brenda	L.	Gruber	11 Ball Hill Rd.	Greenville	PA	16125
511 Ms.	Cheryl		DelVecchio	RD 2 Box 1550	Homer City	PA	15748
512 Ms.	Deborah		Testa	223 North Main St.	Houston	PA	15342
513 Ms.	Dolores	J.	Bracken	525 Bi Road	Julian	PA	16844
514 Mr.	John	W.	Gray	RD 6 Box 279G	Kittanning	PA	16201
515 Ms.	Sherry	A.	Mattern	150 Buffalo Rd. Suite 206	Lewisburg	PA	17837
516Mr.	Larry	G.	Garverick	360 Gerlo Terr. Apt. 5	S. Williamsport	PA	17701
517Mr.	George	P.	Swartz	46 Troy Dr.	Lititz	PA	17543-9419
518 Mr. & Mrs.	Erich & Donna		Toews	1147 Graystone Rd.	Manheim	PA	17545
519Ms.	Carolyn	L.	Bishop	659 Sawmili Rd.	Mechanicsburg	PA	17055
520Mr.	Lewis	G.		441 N. Prince St.	Millersville	PA	17551
521 Ms.	Susan	A.	Elcher	RR 1 Box 158	Monroeton	PA	18832
522 Ms.	Kim	L.	Bailie	RD 1 Box 231-B	New Bethlehem	PA	16242
523 Mr. & Mrs.	Robert & Darlene	<u></u>	Longmore	RR 1 Box 65A	Noxen	PA	18636
524 Dr.	Pete		Watson	P.O. Box 276	Grangeville	PA	17859
525 Mr. & Mrs.	Timothy & Lisa	T	Wehr	3566 Highland Rd.	Orefield	PA	18069
526 Mr.	Randall	D.	Wehr	45 Grelss Rd.	Alburtis	PA	18011
527 Ms.	Miriam	R.	Wehr	4412 Wolfs Crossing Rd.	Orefield	PA	18069
528 Mr.	Donald	E.	Wehr	4412 Wolfs Crossing Rd.	Orefield	PA	18069
529 Mr. & Mrs.	Jerry & Pat	 	Wehr	5560 Ridge Rd.	Slatington	PA	18080
530 Ms.	Holly	W.	Kirk	5522 Homeville Rd.	Oxford	PA	19363
531 Mr.	Stephen	T	Frantz	RD 1 Box 432	Pine Grove	PA	17963
532 Ms.	Ethel		Gaines	HC Box 3	Pottersdale	PA	16871
533 Ms.	Dolores	B.	Skarluk	RD 1 Box 199	Punxsutawney	PA	15767-8653
534 Mr.	Kurt	J.	Schneider	8740 Easton Rd.	Revere	PA	18953
535 Ms.	Tammy		Tillotson	RR 1 Box 155A	Roaring Branch	PA	17765
536 Mr.	Craig		Parker	Box 111	Canton	PA	17724
537 Mr.	Scott	C.	Palmer	RD 2 Box 43A	Canton	PA	17724
538 Ms.	Rebecca	M.	Hill	RD 1 Box 499	Saltsburg	PA	15681
539 Mr.	Ken	1_	Lazaar	119 Queen Esther Dr.	Sayre	PA	18840
540 Mr.	Barry	L.	Moyer	RR 1 Box 13	Scotrum	PA	18355

No Mr/Mrs/Ms		M		師講出建設立置置機構開業		Bath	Zo
541 Ms.	Ruth		Harpster	RD 1 Box 125	Spring Mills	PA	18875
542 Ms.	Jennifer		Sinisi	424 Waupelani Dr. L-21	State College	PA	16801
543 Mr.	Ralph	1	Wingenroth	545 Mt. Airy Rd.	Stevens		17578
544 Ms.	Linda	1	Wideman	RR 1 Box 248	Stillwater		17878
545 Mr.	Philip	E.	Northelmer	RR 1 Box 177A	Trout Run		17771
546 Ms.	Vivian	1=-	Hess	RR 1 Box 103	White Haven	PA	18661
547 Mr.	Gerald	L.	Edwards	RD 1 Box 57	Youngsville		16371
548 Mr.	William	R.	Bearley, Jr.	33 Meadow Ln.	Danville	PA	17821
549 Ms.	Donna		Shrader	137 McKean Rd.	Freeport	PA	16229
550 Mr.	Linda	S.	Sager	721 N. Shippen St. 2nd. Fl. #4	Lancaster	PA	17602
551 Mr.	Daryl	L.	Nace	2324 Chestnut St.	Camp Hill	PA	17011
552 Ms.	Barbara	L.	Hubbard	11918 Hubbard Rd.	Conneaut Lake	PA	16316
553 Ms.	Margaret	T.	Hinterlang	119 Blakely Rd.	Butler	PA	16002
554 Ms.	Mary	A.	DeFlore	3453 Carmela Dr.	New Castle	PA	16105
555Mr.	Blakely		Eyster	1659 Princeton Dr.	State College	PA	16803
558 Mr.	Jeff		Wagner	RR 1 Box 377A	Northumberland	PA	17844
557 Ms.	Kathy	R.	Wagner	RR 1 Box 377A	Northumberland	PA	17844
558 Ms.	Sally	K.	Lewis	638 Queen St.	Northumberland	PA	17844
559 Ms.	Gail		Carroll	431 Market St.	Lewisburg	PA	17837
560 Mr.	Joseph	1	Carroll	431 Market St.	Lewisburg	PA	17837
561 Mr.	Thomas	F.	Hess	RD 3 Box 473	Lewisburg	PA	17837
582 Ms.	Dianne	L.	McCormick	Box 1	Allenwood	PA	17810
563 Ms.	Elaine	G.	Hess	RR 3 Box 473	Lewisburg	PA	17837
564 Ms.	D. Sue		Wells	RD 1 Box 290E	Watsontown	PA	17777
565 Mr.	Thomas	M.	Snoddy	RR 1 Box 36	Allenwood	PA	17810
566 Mr. & Mrs.	Bradley & Anita		Flick	147 W. Sheridan Ave.	Annville	PA	17003
567 Mr.	Robert	B .	Fisher, Jr.	RD 1 Box 1503	New Columbia	PA	17856
568 Mr.	Donald	E.	Aldinger	4410 Wolf's Crossing Rd.	Orefield	PA	18069
569 Mr.	Dale		Copper	RD 6 Box 474	New Castle	PA	16101
570 Ms.	Jessie	G.	Donahue	501 Deer Ridge Rd.	Sarver	PA	16055
571 Ms.	Leona	E.	Sikorski	P.O. Box 293 158 Blakely Rd.	Saxonburg	PA	16056-0293
572 Ms.	Patricia	0.	Offstein	P.O. Box 683	Butler	PA	16003
573 Mr.	Terry	L.	Zwald	402 Burke St.	Jersey Shore	PA	17740
574 Ms.	Terry		Wheeland	RR 1 Box 228	Allenwood	PA	17810
575 Ms.	Cynthia		Gorasi	1421 Washington Blvd.	Williamsport	PA	17701
576 Ms.	Kimberly	A.	Venti	1776 Memorial Ave.	Williamsport	PA	17701

No MANAGEME	曲直ば行び車曲曲	M	MARCY III	斯斯斯·斯斯·斯斯斯斯斯斯斯斯斯斯斯斯斯斯斯斯斯斯斯斯斯斯斯斯斯斯斯斯斯斯	新期限 (2)/開開初	EFET.	R PZD
577 Ms.	Dawn	A.	Banks-Ruck	712 Cherry St.	Williamsport	PA	17701
578 Ms.	Teresa	Α	Snyder	616 Duke St.	Northumberland	PA	17857
579 Mr.	Haroid	M.	Smith, Jr.	700 Boring Bridge Rd.	Lewisberry	PA	17339
580 Ms.	Sandra	L.	Zartler	2730 Blue Ball Rd.	Elkton	MD	21921
581 Mr.	Donald		Edwards	9154 Woodbine Rd.	Airville	PA	17302
582 Ms.	Susan	C.	Brundick	3707 Federal Lane	Abingdon	MD	21009-2743
583 Ms.	Christina	J.	Muller	16 Snowdrift Rd.	Kutztown	PA	19530
584 Ms.	Mary	В.	Marzulli	8340 Walker St.	Philadelphia	PA	19138
585 Mrs.	Maetrel	M.	Yetter	3855 Germantown Pike	Collegeville	PA	19426
586 Ms.	Yda	1	Schreuder	126 Crestwood Rd.	Landenberg	PA	19350
587 Mr.	S.	L.	Slavich	1733 Freida Ave.	Johnstown	PA	15902
588 Mr.	Jay		Catherman	328 E. Siddonsburg Rd.	Mechanicsburg	PA	17055
589 Ms.	Jennifer	L.	Kinsey	170-B Susquehanna St.	Middletown	PA	17057
590Mr.	Edwin	E.	Hollenbach	P.O. Box 117	New Berlin	PA	17855
591 Ms.	Tracy		Baylor	1201 Jefferson Ave.	Lewisburg	PA	17837
592 Ms.	Suzanne		Eves	RD 1 Box 240	Robesonia	PA	19551
593 Ms.	Marilyn	J.	Dale	RR 2 Box 29	Harvey's Lake	PA	18618
594 Mr.	Robert		Dale	196A North Mountain Rd.		PA	18657
595 Mr.	Frank		Zupancic	6501B Brownsville Rd.	Pittsburgh	PA	15236
596 Mr.	Willard	G.	Hutchins	RR 7 Box 7132	Muncy	PA	17756
597 Ms.	Alice	P	Hutchins	RR 7 Box 7132	Muncy	PA	17756
598 Ms.	Mary	L.	Patton	359 Pleasant Dr.	Aliquippa	PA	15001
599 Ms.	Cynthia	A.	Sudor	1205 Ridge Rd.	Grantville	PA	17028
800 Mr.	Charles		Anchor, Jr.	RR 1 Box 81	New Columbia	PA	17856
601 Ms.	Elizabeth	Н.	Smith	RD 1 Box 29	Orangeville	PA	17859
602 Ms.	Sandra	J.	Swope	RD 1 Box 675	Needmore	PA	17238
603 Ms.	Lesley		Hanes	RR1 Box 292	Catawissa	PA	17820
604 Ms.	Vicki	R.	Sicher	5111 Ridge Rd.	Elizabethtown	PA	17022
605 Ms.	Joan	E.	Haire-Camp	RD 1 Box 34	Mifflinburg	PA	17844
606 Ms.	Judi		Spence	RR 2 Box 234	Mifflinburg	PA	17844
607 Ms.	Paula	C.	Hosking	P.O. Box 242	Benton	PA	17814
608 Ms.	Debra	F.	Bums	RD 2 Box 88	Middleburg	PA	17842
609 Ms.	Carrie	L.	Walter	RR 3 Box 187A	Lewisburg	PA	17837
610 Ms.	Kathleen	A.	Flock	RR 3 Box 426	Middleburg	PA	17842
611 Ms.	Paula	S.	Klonaris	1051 Zook Rd.	Atglen	PA	19310
612Mr. & Mrs.	Dennis & Linda		Breanig	175 Deysher Rd.	Fleetwood	PA	19522

ECINCELIAMMENTS	自己通過的計算的計算的開始	11/14	liasa (Plynavi	ha and the control of the same and	[4] [4] [4] [4] [4] [4] [4] [4] [4] [4]	ELICIE	阿斯利斯斯
613 Ms.	Heather						17701
614 Mr.	Thomas	E.	Plummer	4760 Breezy Vista Ln.	York	PA	17406
615Mr.	Harlan	W.	Bower	RR 1 Box 217	Trout Run	PA	17771
616 Ms.	Patricia	A.		RR 1 Box 217	Trout Run	PA	17771
617 Mr.	Frank	L.	Hartley	RR 1 Box 216-A	Trout Run	PA	17771
618Mr.	William	M.	Waters	RR 1 Box 115	Trout Run	PA	17771
619 Mr.	Jay	S.	Hock	RR 4 Box 323A	Bloomsburg	PA	17815
620 Ms.	Elizabeth	W.	Blair	HC 31 Box 298	Williamsport	PA	17701
621 Mr.	James	J.	Pelliccio, Jr.	Box 351 E. White Oak Rd.	Paradise	PA	17562
622 Ms.	Linda	K.	Albinson	102 Willis Rd.	Newville		17241
623 Ms.	Lois				Sunbury		17801]
624 Ms.	Jean		Rine	500 Beaver Run Rd.	Danville	PA	17821
625 Mr.	Matthew		VanZile	441 Sheridan St.	Williamsport	PA	17701
626 Ms.	Nancy	E.	Reader	RD 1 Box 152	Sunbury	PA	17801
627 Ms.	Beth		Snyder	RD 1 Box 151C	Sunbury	PA	17801
628 Mr. & Mrs.	Christopher & Beth		Firme	411 Chestnut St.	Waynesboro	PA	17268
629 Ms.	Carolanne		Craig-Granic	1226 Lord Sterling Rd.	Washington Cross	PA	18977
630 Mr.	Irvin		Lichtenstein	704 Preston Rd.	Erdenheim	PA	19038-7327
631 Mr.	Jack		Martin	217 Sykes Ln.	Wallingford	PA	19086
632 The Honorable	Brett		Feese	House Post Office Box 202020	Harrisburg	PA	17120-2020
633 Mr.	Patrick	M.	O'Connell	121 Oak Hill Dr.	Hanover	PA	17331
634 Ms.	Lois		Morey	RD 1 Box 202	Galeton	PA	16922
635 Mr.	Jeffrey	S.	Weik	2914 Center Rd.	Northampton	PA	18067
636 Ms.	Kate	R.	Clark	202 Burnside St.	Bellefonte	PA	16823
637 Mr.	Gregory	K.	Tothero	3098 Westover Dr.	State College	PA	16803
638 Ms.	Jacqueline	<u> </u>	Warner	328 N. 8th. St.	Lehighton	PA	18235
639 Ms.	Rachel	ļ	Burger	4079 Kistler Rd.	Schnecksville	PA	18078
640 Ms.	Becky Jo	<u> </u>	Megargel	67 Center Ave.	Mt. Pocono	PA	18344-1641
641 Ms.	Dorothy	R.	Toll	1659 Jarrettown Rd.	Dresher	PA	19025
642 Mr.	Robert	IJ.	LaBar	1459 Hemlock Farms	Hawley	PA	18428
643 Mr.	Douglas	W.	Kutz	1221 Bossler Rd.	Elizabethtown	PA	17022-9773
644 Mr.	Steve		Koeppen	3050 Bull Frog Rd.	Taneytown	MD	21787
645 Mr.	John		Abeline	108 Robinson Rd.	Mohawk	NY	13407
646 Mr.	Ronald	1	Comstock	11 King St.	Wellsboro	PA	16901
647 Ms.	Patty		Lamber	2354 W. Gatesburg Rd.	Warriors Mark	PA	16877
648 Ms.	Linda	A.	Bennett	1831 Richmond Ave.	Bethlehem	PA	18018

No. III MININI ZINISI		IM				Bus	IF Zp
649 Mr.	Carl	C.	Demrow	Five Joy St.	Boston	MA	02108
650 Ms.	Deborah	E.	Lynch			PA	
651 Mr.	Randall		Cooley	P.O. Box 8773 RCSOB	Harrisburg	PA	17105-8773
652Ms.	Terry		Brown	229 Greenview Dr.	Lancaster	PA	17601-4997
653 Mr. & Mrs.	Richard & Debra		Sausman	2398 Scenic Dr.	Manheim	PA	17545
654 Ms.	Debra	A.	Carman	2009 Blue Ball Rd.	Elkton	MD	21921
655 Ms.	LaRue	S.	VanZile	2417 N. Hills Dr.	Williamsport	PA	17701
656 Mr. & Mrs.	Bill & Vicki		Smedley	RR 5 Box 525	Jersey Shore	PA	17740
657 Mr. & Mrs.	Herbert & Sylvia	1	Booker	P.O. Box 88	Coudersport	PA	16915-0088
658 Mr. & Mrs.	James & Loretta	†	Berger	RR 3 Box 179A	Coudersport	PA	16915-0088
659 Mr.	Robert	·	Deemer	11 Kelsey St.	Wellsboro	PA	16901
660 Mr.	Chuck	_	Dillon	RR 4 Box 130B	Wellsboro	PA	16901
661 Mr. & Mrs.	Ray & Patricia	- 	Groff	2080 West End Rd.	Bath	PA	18014
662 Mr. & Mrs.	Dorsey & Ethel	 	Musselman	RR 2 Box 704	Bedford	PA	15522
663 Ms.	Deborah	S.	Grove	RD 1 Box 483	Petersburg	PA	16669
664 Ms.	Joanne	1	Matthews	P.O. Box 731	Ligonier	PA	15658
665 Ms.	Brenda	+	Petroski	86 Jamison Dr.	York	PA	17402
666 Mr. & Mrs.	Nancy & Joe	+-	Robinson	RD 1 Box 273	Seven Valleys	PA	17360
667 Mr.	D. A.	 	Truby	850 Beech St. apt. 602	Washington	PA	15301
668 Mr.	Paul	1-	Truby	850 Beech St. apt. 602	Washington	PA	15301
669 Ms.	Donna	J.	Bakaitis	25 Chambers Hill Rd.	Washington	PA	15301
670 Mr.	Preston		Patterson	41 Moore Ave.	Washington	PA	15301
671 Ms.	Virginia	R.	Yorke	127 Richland Ave.	Canonsburg	PA	15317
672 Ms.	Susan	E.	Patterson	41 Moore Ave.	Washington	PA	15301
673 Mr.	Edward	P.	Yorke, Jr.	249 Glenn Ave.	Canonsburg	PA	15317
674 Ms.	Cammi	J.	Yorke	249 Glenn Ave.	Canonsburg	PA	15317
675Mr.	David	В.	Eyster	18 Casey Dr.	Williamsport	PA	17701
676 Ms.	Patricia	M.	Eyster	1659 Princeton Dr.	State College	PA	16803
677 Ms.	Linda	S.	Gundrum	RD 1 Box 390	Pine Grove	PA	17963
678 Mr.	George	C.	Gundrum	RD 1 Box 390	Pine Grove	PA	17963
679 Ms.	Angela	J.	Byme	10238 Reeder Rd.	Mercersburg	PA	17236
680Ms.	Mary	+	Long	1020 LWW	St. Thomas	PA	17236
681 Ms.	Alvie	 	May	1020 LWW	St. Thomas	PA	17236
682 Mr.	Demy		Krause	10238 1/2 Reeder Rd.	Mercersburg	PA	17236
683 Ms.	Midge		Tressler	10238 Reeder Rd.	Mercersburg	PA	17236
684 Ms.	Leah		Strouphauer	RD 1 Box 432	Pine Grove	PA	17963

IN THE WAY	11 重量 [1] [1]	M		制機構調整にごごう注意機構		State	11 210 ×
685 Mr.	Woody		Fritz	12 Mifflin St.	Pine Grove		17963
686 Mr.	Brian		Beck	827 Church St.	Lebanon	PA	17046
687 Ms.	Judy		Tomb	RD 2 Box 577	Pine Grove	PA	17963
688 Mr.	Barry		Tobias	19 W. Mill St.	Pine Grove	PA	17963
689 Ms.	Jean	E.	Tobias	19 W. Mill St.	Pine Grove		17963
690 Mr.	Ronald		Tobias	19 W. Mill St.	Pine Grove		17963
691 Ms.	Kade	C.	Haldeman	Main St.	<u> </u>		17963
692 Ms.	Haley	L.	Haldeman	186 N. Tulpehocken St.	Pine Grove	PA	17963
693 Mr.	John	R.	Haldeman	RD 1 Box 528-C	Pine Grove	PA	17963
694 Mr.	Luke		Schnoke	RD 5	Pine Grove	PA	17963
695 Mr.	Ronald	E.	Klinger	RD 1 Box 492	Pine Grove	PA	17963
696 Mr.	Jon	M.	Seiger	65 Mifflin St.	Pine Grove	PA	17963
697 Mr.	Ricky	L.	Krause	11 Brookside Ct.	Pine Grove	PA	17963
698 Ms.	Linda		Leese	207 Main St.	Joliett	PA	17981
699 Ms.	Marcle		Kozich	RD 1 Box 82	Tower City	PA	17980
700 Ms.	Tina	L.	Felty	RD 1 Box 250A	Pine Grove	PA	17963
701 Mr.	Shaun	L.	Williams	RD 1 Box 250A	Pine Grove	PA	17963
702 Mr.	Michael		Pesta	RD 2 Box 4532	Jonestown	PA	17038
703 Ms.	Kathy	1	Walsh	RD 2 Box 803	Schuylkill Haven	PA	17972
704 Mr.	Rick	 	Glay	RD 2 Box 4532	Jonestown	PA	17038
705 Ms.	Dana	L.	Williams	RD 4 Box 276	Pine Grove	PA	17963
706 Ms.	Peg	=-	Klinger	RD 1 Box 492	Pine Grove	PA	17963
707 Mr.	Edward	G.	Ditzler	RD 1 Box 380	Pine Grove	PA	17963
708 Ms.	Barbara	L.	Rush	RD 1 Box 216	Jefferson	PA	15344
709 Ms.	Donna	1	Walker	2408 Mill Grove Rd.	Pittsburgh	PA	15241
710 Mr.	Kevin	W.	Barnhart	654 Bowlby St.	Waynesburg	PA	15370
711 Ms.	Elizabeth		Hanko	RD 1Box 683	Rices Landing	PA	15357
712 Ms.	Dawn	L.	Wise	207A East South St.	Carmichaels	PA	15320
713 Mr.	Jason	F.	Wright	RD 1 Box 122	Graysville	PA	15337
714 Ms.	Davetta	L.	Bamhart	RR 1 Box 125-D	Carmichaels	PA	15320
715 Mr.	Harry	B.	Kelley, Jr.	RD 3 Box 229W	Waynesburg	PA	15370
716 Mr.	Matthew	J.	McMichael	RD 3 Box 150	Smithfield	PA	15478
717 Mr.	Wallace		Filiatrault	398 N. Porter St.	Waynesburg	PA	15370
718 Mr.	John	T	McCartney	P.O. Box 12	Mather	PA	15346
719 Mr.	Marvin	D.	Knight	P.O. Box 267	Jefferson	PA	15344
720 Mr.	Brian	L	Mason	RD 1 Box 129	Aleppo	PA	15310-9612

No MriMesiMs	FN	M		計劃的數學的數學的數學的數學的	問題は出る政権制制		II Zio
721 Ms.	Patricia		Smitley	22 Circle St.	Mather	PA	15346
722 Ms.	Carrie		Reeves	P.O. Box 376	Clarksville	PA	15322
723 Ms.	Shirley		Bumfleld	RD 1 Box 63-B	Waynesburg	PA	15370
724 Ms.	Lovie		Coffman	RD 1 Box 25C	Sycamore	PA	15364
725 Mr.	Cary		Hoge	RD 4 Box 266A	Waynesburg	PA	15370
726 Ms.	Brenda	J.	Wagner	36 2nd. St.	Shillington	PA	19607
727 Mr.	Glen		Hartman	207 Charles Blvd.	Sinking Spring	PA	19608
728 Ms.	Sharon		Meischeid	825 N. 38th. St.	Alientown	PA	18104
729 Ms.	Linda	L.	Ganter	1055 Strause Ln.	Leesport	PA	19533
730 Ms.	Deborah		Ryan	555 Friendship Dr.	Honeybrook	PA	19544
731 Ms.	Candie		Heller	521 Margaret St.	Reading	PA	19611
732 Ms.	Jolyn		George	461 Texter Mt. Rd.	Robesonia	PA	19551
733 Ms.	Carmen		Holland	2905 Merritt Parkway	Sinking Spring	PA	19608
734 Mr.	Clinton	W.	Kendall	293 Cushion Peak Rd.	Reinholds	PA	17569
735Ms.	Ann	L.	Leinbach	2215 Perkiomen Ave.	Reading	PA	19606
736 Mr.	David	<u> </u>	Hershey	174 Vinemont Rd.	Reinholds	PA	17569
737 Mr.	David	 		460 Wegnan Rd.	Reading	PA	19606
738 Ms.	Christine	1			Reading	PA	19606
739 Ms.	Kathy	1	Stump	101 Monroe St.	Denver	PA	17517
740 Mr.	Robert	L.	1	25 Hollow Rd.	Reinholds	PA	17569
741 Ms.	Nicola	1	Tilley	509 Martins Rd.	Sinking Spring	PA	19608
742 Ms.	Jennifer	A.	Martin	2809 Marietta Ave.	Lancaster	PA	17601
743 Ms.	Carol		Heckman	RR 2 Box 3500	Wernersville	PA	19585
744 Mr.	Kent	W.	Yanchek	RR 2 Box 3500	Wernersville	PA	19565
745 Ms.	Sharon		Oxenreider	RD1 Box 412A	Womelsdorf	PA	19567
746 Ms.	Rebecca		Oxenrelder	RD1 Box 412A	Womelsdorf	PA	19567
747 Ms.	Beverly		Lash	350 N. Galen Hall Rd.	Wemersville	PA	19565
748 Ms.	Keri	L.	Fisher	17 W. Penn Ave.	Robesonia	PA	19551
749 Ms.	Mary		Hower	5821 Fourt Point Rd.	Bethel	PA	19507
750 Mr.	James	1	Cramp	73 Mountain Top Rd.	Wemersville	PA	19565
751 Mr.	Richard	1	Bromley	2545 Cumberland Ave.	Mt. Penn	PA	19606
752 Ms.	Kerry	C.	Fisher	774 Fritztown Rd.	Sinking Spring	PA	19608
753 Mr.	Richard	1	Dalton	1403 Linden St.	Reading	PA	19604
754 Mr.	Earl	R.	Burg	134 W. Broadway	Red Lion	PA	17356
755 Ms.	Dawn	M.	Burg	134 W. Broadway	Red Lion	PA	17356
756 Ms.	Patricia		Laird	RD 3 Box 397	Dallastown	PA	17313

		M					Z
757 Ms.	Bettina	K.	Weinhold	1211 Whitfield Blvd.	West Lawn	PA	19609
758 Ms.	Evelyn		Swalm	725 Ruth St.	Sinking Spring	PA	19608
759 Ms.	Mary	R.	Shirk	273 Boottalk Rd.	Narvon	PA	17555
760 Mr.	Andrew	B.	Galligan	2078 Main St.	Narvon	PA	17555
761 Ms.	Mary	A.	Galligan	2078 Main St.	Narvon	PA	17555
762 Ms.	Victoria	G.	Sher	P.O. Box 59	Oley	PA	19547
763 Ms.	Susanne	L.	Ott	7288 Camp Meeting Rd.	New Tipoli	PA	18066
764 Ms.	Deirdre		Rahn	311 Berne Rd.	Hamburg	PA	19526
765 Ms.	Anne	M.	Woodring	520 Snyder Rd.	West Lawn	PA	19609
786 Mr.	Donald	M.	Woodring	520 Snyder Rd.	West Lawn	PA	19609
767 Ms.	Cari		Nelso	226 S. 4th. St.	Womelsdorf	PA	19567
768 Ms.	Pat		Manar	40 Water St.	Oley	PA	19547
769 Ms.	Mary	E.	Blair	2607 Dalin Dr.	West Lawn	PA	19609
770 Mr.	James	M.	Rhodenizer	1114 Locust St.	Reading	PA	19604
771 Mr.	Norman	K.	Fryberger	17 N. Church St.	Mohnton	PA	19540
772 Ms.	Katherine		Rightmyer	1036 Tilden Rd.	Mohrsville	PA	19541
773 Ms.	Sharon		Permine	1010 Syner Rd.	Annville	PA	17003
774 Ms.	Abby		Spicer	1010 Syner Rd.	Annville	PA	17003
775 Ms.	Maria	T.	Biancone	17 N. Church St.	Mohnton	PA	19540-170
776 Mr.	Roger	L.	Behn, Sr.	592 Blandon Rd.	Fleetwood	PA	19522
777 Mr.	Paul	R.	Riffle	22 Cralou Dr.	Fleetwood	PA	19522
778 Ms.	Paula		Horning	60 Clearview Dr.	Reinholds	PA	17569
779 Mr.	Punch		Haring	858 Christmas Village Rd.	Bernville	PA	19506
780 Ms.	Susanne		Flori	17 Water St.	Oley	PA	19547
781 Ms.	Zita	M.	Codi	312 Miller	Sinking Spring	PA	19608
782 Ms.	Marty		Sheffer	12 Russian Olive Dr.	Etters	PA	17319
783 Ms.	Cindy	L.	Boyer	17 Water St.	Oley	PA	19547
784 Ms.	Audrey		Nace	1502 Walnut St.	Lansdale	PA	19446
785 Mr.	David	C.	Nace	1502 Walnut St.	Lansdale	PA	19446
786 Ms.	Marie	S.	Brown	P.O. Box 186	Grantville	PA	17028
787 Ms.	Marilyn		Trusky	RD 3	Jonestown	PA	17038
788 Mr.	Charles		Whitaker	RD 1 Box 1265	Mohnton	PA	19540
789 Ms.	Dawn	R.	Berk	RD 3 Box 283A	Kutztown	PA	19530
790 Mr.	Guy	D.	Geschwindt	428 Oak St.	Port Carbon	PA	17965
791 Ms.	Janet	M.	Geschwindt	428 Oak St.	Port Carbon	PA	17965
792 Ms.	Vickie		Kauffman	435 Oak St.	Johnstown	PA	15902

	国际 中央区域区域 基本	WI		開音的は記載は進業等は同じまって、「をはず」			
793 Ms.	Debra	H.	Damarodis	900 Newportville Rd.	Croydon	PA	19021
794 Mr.	Craig		Carpenter	106 Glen Mawr Dr.	Ambler	PA	19002
795 Ms.	Sandra	V.	Kelly	361 Ivy Mills Rd.	Glen Mills	PA	19342
796 Ms.	Judy		Scott	325 Noble Rd.	Atgien	PA	19310
797 Mr.	Steve		Reich	2069 Riverside Dr.	S. Williamsport	PA	17701
798 Ms.	Judy		Hower	RR 1, Box 165	Alienwood	PA	17810
799 Mr.	J. William		Housel	RR 5, Box 490	Jersey Shore	PA	17740
800 Ms.	Rachel	A.	Bechdel	138 Hillcrest St.	Howard	PA	16841
801 Mr.	Istran		Merchenthol	806 Byers Rd	Chester Springs	PA	19425
802 Ms.	Mindy		Pollock	395 F. Madrid Avenue	Bloomsburg	PA	17815
803 Mrs.	Brenda	L.	Rizzo	RR 2, Box 472	Sugarloaf	PA	18249
804 Ms.	Margaret	C.	Riggin	9140 Hickory Dr.	Oxford	PA	19363
805 Mr.	Curt		Althouse	1165 Kramer Mill Rd	Denver	PA	17517
806 Mr.	Brian	P.	Haas	HC 31, Box 240	Williamsport	PA	17701
807 Ms.	Karen		Insley	P.O. Box 226	Woolrich	PA	17779
808 Ms.	Doris		Wallace	185 School House Rd.	Boyertown	PA	19512
809 Ms.	Beatrice		isola	RD 1, Box 511	Petersburg	PA	16669
810 Ms.	Jenna		Miller	3646 Lil. Wolf Circle	Orefield	PA	18069
811 Ms.	Phyllisjo		Dempsey	1397 Pine Ridge	Bushkill	PA	18324
812 Mr.	Alten	C.	Weaver	RR 1, Box 414	Loganton	PA	17747
813 Ms.	Denise		Walker	RR 2, Box 354	Canton	PA	17724
814 Ms.	Ashley		Weaver	RR 1, Box 414	Loganton	PA	17747
815 Ms.	Donna	R.	Weaver	RR 1, Box 414	Loganton	PA	17747
816 Mr.	Donald		Kline	RR 2, Box 12	Montgomery	PA	17752
817 Mr.	Harry				Montgomery	PA	17752
818 Ms.	Judith	Α.	Kauffman	3340 Cape Horn Rd.	Red Lion	PA	17356
819 Mr.	Ronald	W.	Harrison	240 Essex Rd.	Warminster	PA	18974
820 Mr.	Edward	C.	McCormick	110 Pony Trail Lane	Renfrew	PA	16053
821 Ms.	Virginia	W.	Kelsall	1220 Kelsall Rd.	Coatesville	PA	19320
822 Ms.	Holly	S.	Orr	7163 Ferry Rd.	New Hope	PA	18938
823 Ms.	Joyce	F	Schroder	6596 Groveland Rd.	Pipersville	PA	18947
824 Mr.	M.	L.	Surgma	136 State St.	St. Mary's	PA	15857
825 Ms.	Judith	G.	Abahams	504 Tioga St.	Johnstown	PA	15905
826 Mr.	Edmond		Jacoby	3518 New Germany	Ebensburg	PA	15931
827 Ms.	Susan		DeLong	RD 2, Box 12	Blairsville	PA	15717
828 Mr.	Terry		Kane	RR 3, Box 46	New Florence	PA	15944

NO BELLIANDE						State	Zb .
829 Mr. & Mrs.	Dorsey			RR 2, Box 704	Bedford	PA	15522
830 Dr.	Frank		Granich	67 Sherman Rd.	Ottsville	PA	18942
831 Mr.	Richard	L.	Tevis	1205 South Market St.	Bloomsburg	PA	17815
832 Mr. & Mrs.	James	L.	Wonderly	RD 1, Box 274 J.	Weedville	PA	15868
833Ms.	Mary Ellen		Shivetts	8240 Fox Ridge Rd.	Pittsburgh	PA	15237
834 Mr. & Mrs.	Roger & Marsha &		Beck	RD 1	Weedville		15868
835 Mr.	Howard			RR 1, Box 381 A	Milton		17847
836 Ms.	Susan		Denson	804 Mt. Zion Dr.			17821
837 Ms.	Janet			RR 1, Box 1350	Forksville		18616
838 Ms.	Gavie			RD 3, Box 172	Tamaqua		18252
839Mr.	William		Helm	335 Orchard Rd.	Mohnton		19540
840 Ms.	Carol	† ·		31 Charlestown Rd.	Washington Boro.	1	17582
841 Ms.	Cynthia	D	Stadel	2370 Fruitville Pike	Lancaster	· · · · · · · · · · · · · · · · · · ·	17601
842 Mr. & Mrs.	Richard		Sausman	2398 Scenic Dr.	Manheim	1	17545
843 Mr. & Mrs.	Robert	 	Hummel	45 Mohns Hill Rd.	Reinholds	PA	17569
844 Ms.	Barbara	A.	Lucia	500 S. Front St. Lot 4	Wrightsville	PA	17368
845 Mr.	David	C.	Rintz	43 B. S. Charlotte St.	Manheim	PA	17545
846 Ms.	Judith	A.	Coover	346 Snyder School Rd.	Bernville	PA	19506
847 Mrs.	Mariene		Sweigart	755 South Ridge Rd.	Denver	PA	17517
848 Ms.	Vicki	L.	Gardner	328 Vinemont Rd.	Reinholds	PA	17569
849Ms.	Kathy	K.	Borowski	151 Icedale Rd.	Honey Brook	PA	19344
850 Mr.	Paul	W.	Wissler	503 Front St.	Lititz	PA	17543
851 Ms.	Sylvia		Grisez	10 Belmont Dr.	Warren	PA	16365
852 Ms.	Lorraine	 	Healey	93 Cedarwood Dr.	Loften	PA	18702
853 Ms.	Mary	†	Pitzer	1368 Spreading Oil	Pittsburgh	PA	15220
854 Mr.	Gene	+-	Brown	P.O. Box 570	Boalsburg	PA	16827
855 Mr.	J. R.	t^-	Fagan	2001 Elmerton Ave.	Harrisburg	PA	17110
856 Mr.	James	E.	Hooper	RD 2, Box 51	Wrightsville	PA	17378
857 Ms.	Beth		Lorensberg	18 School St.	Galeton	PA	16922
858 Mr.	Bob		Taetzsch	1221 N. New St.	Bethlehem	PA	18018
859 Ms.	Esther	G.	Heckman	5121 Pontiac Rd.	Drexel Hill	PA	19026
860 Ms.	Patricia	1_	Williams	8442 Each Leaf Ct.	Columbia	MD	21045
861 Mr.	Horace	R.	Hosbach	737 Amosland Rd.	Morton	PA	19070
862 Ms.	Ginny	ļ	Owen	9211 Flower Ave.	Silver Springs	MD	20901
863 Mr.	John		Clark	4149 Rt. 19	Belmont	NY	14813
884 Mr.	Hudson	A.	Close	307 Broadway	Elmira	NY	14904

NEW MANAGEME	并限的全国和当时的高级	LY13	TELL HELDER	推出。2015年,在1015次。建2世代別開始東	1年至今1月17日。	i lui	關它的影響
865 Mr.	Jim		Owen	9211 Flower Ave.	Silver Spring		20901
866 Ms.	Kelly		Phillips	43 S. Manheim St.	Annville	PA	17003
867 Mr.	Richard		Hiemenz	Kinzua Rd.	Warren	PA	16365
868 Ms.	Debble		LaBerge	5000 Homeville Rd.	Cochranville	PA	19330
869 Mr.	Floyd		Lounsbury	18 School St.	Galeton	PA	16922
870 Mr.	Edward		Zukauskas	108 Washington Ave.	Exeter	PA	18643-1412
871 The Honorable		M.	Perzel	110 Main Capitol Bidg.	Harrisburg	PA	17120-2020
872 Mr.	Mike		Benyo	12 Fawn Dr.	W. Hazleton	PA	18201
873 The Honorable	Rod	E.	Witt	House Post Office Box 202020	Harrisburg	PA	17120-2020
874 Mr.	John		Motz	RR 8 Box 8142A	Stroudsburg	PA	18360
875 Ms.	Margaret	S.	Price	RD 6 Box 6552	Stroudsburg	PA	18360
876 Ms.	Kristy		Green	402 Market St.	Kittanning	PA	16201
877 Ms.	Vivian	J.	Leimseider	117 S. Pennell Rd.	Lima	PA	19037
878 Ms.	Mary	A.	Nissley	66 Newville Rd.	Chalfont	PA	18914
879 Mrs.	Mary	L.	Wenger	24 Gardner Rd.	Duncannon	PA	17020-9767
880 Ms.	Hildy	F.	Rauch	961 Westfield Rd.	Springfield	PA	19064-3834
881 Mr.	Tony		Morell	2011 2nd. Ave.	Lake Ariel	PA	18436-9803
882 Mr.	William		Miehle	1240 Steel Rd.	Havertown	PA	19083
883 Mr.	Jack		Hiehman	5121 Pontiac Rd	Drexel Hill	PA	19026
884 Mr.	Carl	E.	Lorence	3812 Griffin Ln.	Harrisburg	PA	17110-3007
885 Mr.	Joseph	J.	Cushner	3204 Clearview Ave.	Baltimore	MD	21234
886 Mr.	Charles		Hem	Rt 3 Box 7055	Jonestown	PA	17038
887 Ms.	Suzanne		Ahlberg	1289 Church Rd.	Ontanna	PA	17353
888 Mr.	W. Franklin		Bohn	2044 Chestnut St.	Harrisburg	PA	17104
889 Mr. & Mrs.	Jim & Barbara		Beshore	RR 1 Box 54	Bloomsburg	PA	17815-9774
890 Ms.	Pat		Childs	3454 West Dr.	Wellsville	NY	14895
891 Mr.	Ted		Sanderson	6029 Loreley Beach Rd.	White Marsch	MD	21162
892 Ms.	Frances		Jacobi	71 Superior Rd.	Rochester	NY	14625
893 Mr.	Jim		Mayer	328 Rockingham St.	Rochester	PA	14625
894 Mr. & Mrs.	William & Carolyn		Bauer	22 Parkwind Ct.	Baltimore	MD	21234
895Ms.	Charlotte	S.	Popovich	RR 1 Box 303	Mill Hall	PA	17751
896 Mr. & Mrs.	Lee		Fleishman	825 Corbett Rd.	Monkton	MD	21111
897 Mr. & Mrs.	Eric	T	Pedersen	241 Rader School Rd.	Renfrew	PA	16053
898 Mr. & Mrs.	Ralph	T	Seeley	2025 Half Moon Valley Rd.	Port Matilda	PA	16870
899 Mr.	Jerry	Τ	Hirko	P.O. Box 69	Saint Johns	PA	18247
900 Mr. & Mrs.	Ralph	D.	Hoffman	627 Stafford Ave.	Scranton	PA	18505

No all Ministry		MIX				Site	120
901 Ms.	Shirley		Raisch	233 Hughes St.	Williamsport	PA	17701
902 Mr.	Bob		Bernhardy	193 Greenbrook Dr.	Levittown	PA	19055
903 Mr.	Edward	J.	Beck	P.O. Box 251	Cogan Station	PA	17728
904 Mr. & Mrs.	Robert			RR 1, Box 280	Muncy		17756
905 Ms.	Joan			P.O. Box 7713	Lancaster		17604
906 Mr.	John			RR 2, Box 151	Wapwallopen		18660
907 Ms.	Bernadine			RR 2, Box 151	Wapwallopen	PA	18660
908 Mr.	Jeffrey	C.		434 State St.	Lancaster	PA	17603
909 Mr.	Conrad			P.O. Box 7922	Lancaster	PA	17604
910 Mr.	Ted		Flowers	522 Pike St.	Reading	PA	19601
911 Mr. & Mrs.	James		Leaman	416 Roosevelt Ave.	York	PA	17404
912Mr.	Thomas	E.	Burks	290 Franklintown Rd.	Dillsburg	PA	17019
913 Mr.	Raymond	E.	Buck	Box 248	Ulysses	PA	16948
914 Ms.	Irvette	K.		2631 Colebrook Rd.	Elizabethtown	PA	17022
915Mr.	R.		Brodzinski	P.O. Box 315	West Milton	PA	17885
916 Mr. & Mrs.	Thomas	A.	Randali	986 Tourist Park Rd.	Halifax	PA	17032
917 Ms.	Christi		Shultz	RR 3, Box 53	Benton	PA	17814
918 Mr. & Mrs.	William	E.		RR 1, Box 275	Millville	PA	17846
919 Ms.	Mary	L.	Wenger	24 Gardner Rd.	Duncannon	PA	17020
920 Ms.	Rita		Bowman	1090K Ville Rd.	Stevens	PA	17578
921 Mr. & Mrs.	Don	1	Torrey	HC 1, Box 69	Shinglehouse	PA	16748
922 Ms.	Sarita	L.	Welmer	46 Chestnut Run	Elizabethtown	PA	17022
923 Mr.	Haroldean		Pentz	RD 1, Box 15	East Waterford	PA	17021
924 Mr.	J. Ame	1	Fomey	11 Skyview Ln.	Lititz	PA	17543
925 Mr.	Bill	1	Morey	RR 1 Box 202	Galeton	PA	16922
926 Ms.	Louise	1	Hum	2018 Robindale Ave.	Lancaster	PA	17601
927 Ms.	Bonita	1	Wauls	78 N. Third St.	Chambersburg	PA	17201
928 Mr. & Mrs.	J. W.	†	Amberg	5024 N. Convent Ln. apt. G	Philadelphia	PA	19114-3135
929 Mr.	Bob		Tait	P.O. Box 2968	Butler	PA	16003-2968
930 Ms.	Nancy		Megargel	67 Center Ave.	Mt. Pocono	PA	18344-1641
931 Mr.	Leon	E.		220 Yocumtown Rd.	Etters	PA	17319
932 Ms.	Janice		Slaybaugh	P.O. Box 395	Boiling Springs	PA	17007-0395
933 Mr.	James	S.	Straub	RR 1 Box 284	Winfield	PA	17889
934 Mr.	Philip			245 Ellen Ave.	State College	PA	16801
935 Mr.	Charles		Irvin	11540 Simmons Ave.	Taneytown	MD	21787
936 Mr.	Martin	1	Sussman	500 Hilton Ave.	Baltimore	MD	21228

INCH HE LEVEL TO THE LEVEL HE SEED TO THE LEVEL HE	建油油汽油油	HXII		祖祖祖祖(1777年) 建建		360	10年25世
937 Mr.	Steve		Williams	8442 Each Leaf Ct.	Columbia		21045-5628
938 The Honorable	John	E.	Barley	House Box 202020	Harrisburg	PA	17120-2020
939 Mr. & Mrs.	Robert & Anne		Magar	545 West Rd.	Webster	NY	14580-1130
940 Ms.	Carol	A.	Howeli	196 Union Church Rd.	Mars	PA	16046
941 Ms.	Connie		Standfast	P.O. Box 224	Reeders	PA	18352
942 Mr.	Joe		Nachtwey	6452 Constitution Rd.	Schnecksville	PA	18078
943 Mr.	Wayne		Gross	RR 1 Box 358	Cresco	PA .	18326
944 Ms.	Solvejj		Wills	47 stones Throw	E. Stroudsburg	PA	18301
945 Ms.	Carol		Easter	109 Traugers Crossing Rd.	Kintnersville	PA	18930
946 Mr.	David		Angst	6715 Memorial Rd.	New Tripoli	PA	18066
947 Ms.	Florence	Z.	Weber	560 Sheridan Mnr. #D-8	Wilkes Barre	PA	18702-6233
948Mr.	David	A.	Kreider	RR 4 Box 125A	Bloomsburg	PA	17815
949 Ms.	Selly	E.	Kuhns	RR 4 Box 436	Montoursville	PA	17754
950Mr. & Mrs.	David & Betty Jo	1	Paris	RR 1 Box 667	Newmanstown	PA	17073
951 Ms.	Carol	L.	Sprecher	310 W. Fulton St.	New Holland	PA	17557
952Mr.	Thomas	D.	Tulli	1015 Kreps Rd.	Lancaster	PA	17603
953 Ms.	Karen	L.	Stottzfus	404 E. Oregon Rd.	Lititz	PA	17543
954 Ms.	Shelley		Dering	1020 Olde Hickory Rd.	Lancaster	PA	17601
955 Ms.	Carole	1	Shirk	101 S. Groffdale Rd.	Leola	PA	17540
956 Ms.	Shella	L.	Wallick	227 Valley View Rd.	York	PA	17402
957 Mr.	Charles	R.	Wessner	372 Snyder School Rd.	Bemville	PA	19506
958 Mr.	Brian	A.	Feick	1904 Van Reed Rd. apt. G3	Wyomissing	PA	19610
95 9 M s.	Jan	S.	Kaufman	305 Mail Route Rd.	Reading	PA	19608
960 Mr.	Walter	A.	Spatz	108 N. Centre Ave.	Leesport	PA	19533-0375
961 Mr.	Richard	A.	Kaufman	305 Mail Route Rd.	Reading	PA	19608
962 Mr. & Mrs.	Kenneth & Judith	↓	Cottrell	RD 5 Box 5281	Moscow	PA	18444
963 Ms.	Ida	M.	Moore	1179 Bridgeton Rd.	Fawn Grove	PA	17321
964 Ms.	Cynthia	Α.	Campbell	2129 Colebrook Rd.	Lebanon	PA	17042
965 Ms.	Nancy	R.	Hilferty	3280 E. Prospect Rd.	York	PA	17402-8682
966 Ms.	Lori	Α.	Lorson	2664 Pleasant View Rd.	Cogan Station	PA	17728
967 Ms.	Melissa		Lohman	441 Sheridan St.	Williamsport	PA	17701
968 Mr.	Andrew		VanZile	2417 N. Hills Dr.	Williamsport	PA	17701
969 Ms.	Carol	D.	VanZile	2417 N. Hills Dr.	Williamsport	PA	17701
970 Ms.	Lori		Lesko	P.O. Box 35	Slate Run	PA	17769
971 Mr. & Mrs.	John & Doris		Dixon	1900A Ravine Rd. apt. 220	Williamsport	PA	17701
972 Ms.	Sherry	L.	Persun	RR 1 Box 335	Cogan Station	PA	17728

MINISTER MENT	(地湾)地區(江北湖區)			建設工作工作工作工作工作工作工作工作工作工作工作工作工作工作工作工作工作工作工作	期間はほどの動態	8 Cabe	E 1920
973 Ms.	Nedra		Cox	RR 2 Box 174-B	Lock Haven		17745
974 Ms.	Holly	M.	Eyster	18 Casey Dr.	Williamsport	PA	17701
975 Mr.	Steven	K.	Persun	RR 1 Box 335	Cogan Station	PA	17728
976 Mr.	Leon	R.	Snyder	RD 1 Box 6	Turbotville	PA	17772
977 Ms.	Shirley	 		RD 1 Box 6	Turbotville	PA	17772
978 Mr.	Christopher	K.	Hughes	29 Arch St.	Montoursville		17754
979Ms.	Evelyn		Hammer	Box 332	North East		16428
980 Ms.	Cynthia			262 Zoskey Rd.	Lilly	PA	15938
981 Ms.	Jeanette			RD 5 Box 210B	Tyrone	PA	16686
982 Ms.	Debra			RD 4 Box 30	Altoona	PA	16601
983 Ms.	Judith		Yates	RD 2 Box 269	Tyrone	PA	16686
984 Ms.	Sidonia		Uzzeli	P.O. Box 103	Snowshoe	PA	16874
985Ms.	George	R.	Uzzeli	P.O. Box 103	Snowshoe	PA	16874
986 Mr.	James	R.	Yates	RD 2 Box 269	Tyrone	PA	16686
987 Ms.	Candace	A.	Lindberg	525 W. Foster Ave. apt. 18	State College	PA	16801
988 Mr.	Francis	G.	 	2034 Meade Ln.	Altoona	PA	16602
989 Ms.	Audrey	1.		2034 Meade Ln.	Altoona	PA	16602
990 Ms.	Judith	M.	Feller	RD 2 Box 2045	Saylorsburg	PA	18353
991 Ms.	Terrie	1	Haney	RD 2 Box 431A	Tyrone	PA	16686
992 Mr.	Mark	1	Baker	RD 2 Box 431A	Tyrone	PA	16686
993 Mr.	Jeffrey	W.	Hines	RD 2 Box 431A	Tyrone	PA	16686
994 Mr.	Michael	+	Baker	RD 2 Box 431A	Tyrone	PA	16686
995 Mr.	Robert	S.	Ramer, Jr.	59 Bradford Dr.	Leola	PA	17540-1702
996 Ms.	Dorothy	S.	Hunter	RD 2	Ligonier	PA	15658
997 Mr.	Thomas	M.	McEwen	225 Sunset Dr.	Beaver Falls	PA	15010
998 Mrs.	Brook	1		i 825 S. Ridge Rd.	Denver	PA	17517
999 Ms.	Rebecca	L.	Dougherty	815 Frosty Hill Rd.	Airville	PA	17502
1000Ms.	Gloria	 	Schwartz	104 E. Meadow St.	Shavertown	PA	18708
1001 Ms.	Anne	W.	Ward	RD 3 Box 244	Troy	PA	16947
1002 Mr.	Robert	1	Trautwein	1266 McConnell Dr.	Williamsport	PA	17701
1003Ms.	Christie	†-	Fortin		Cogan Station	PA	17728
1004 Ms.	Jan	1	Kirby	1233 W. 8th. Ave.	S. Williamsport	PA	17701
1005 Mr.	Creon	В.	Frederickso		Williamsport	PA	17701
1006Mr.	Josh	Ť	Bostley	259 Broad St.	Montoursville	PA	17754
1007 Ms.	Marty		Wolfgang	158 E. Water St.	Muncy	PA	17756
1008 Mr.	David	T	Sees	1360 Mansel Ave.	Williamsport	PA	17701

Name of the last o		 دُ				Same	11175
1009 Mr.	Kenneth	Ì	, Julck	RD 1 Box 1015	Nescopeck	PA	18635
1010 Ms.	Joan	E.	Heintzelman	138 Mahoning Heights Rd.	Lehighton	PA	18235
1011 Ms.	Sharon			RR 1 Box 251D	Stillwater	PA	17878
1012 Ms.						PΛ	17922
1013 Ms.	Olive	ፉ.	,Killmeyer -	233 Lively Rd.	Eighty Four	PA	15330
1014 Ms.	Alice			201 Gibralter Dr.	Pittsburgh	PA	15239
1015Mr.	Harry	K.	Frye	965 Tuscarawas Rd.	Beaver	PA	15009
1016 Mr.	Andrew		Miller	RD 3 Church Ln.	Allison Park	PA	15101
1017 Ms.	Samantha		Rossey	RD 1Box 502	Venus	PA	16364
1018 Ms.	Kathy		Stewart	2639 Castle Dr.	Bensalem	PA	19020
1019 Mr.	Timothy	L.	Panik	206 Overlook Dr.	Coatesville	PA	19320
1020 Mr. & Mrs.	Ralph & Anne		Ruth	1861 Tower Rd.	Mohrsville	PA	19541
1021 Mr. & Mrs.	John & Alexis		Hyneman	RR 1Box 393	Birdsboro	PA	19508
1022 Mr.	Thomas	W.	Zborel	2395 Wehr Mill Rd.	Allentown	PA	18104
1023 Mr.	Charles	R.	Beer	130 Spruce Hollow Rd.	Palmerton	PA	18071
1024 Mr. & Mrs.	Raymond & Eileen		Prussman	424 Jefferson St.	Oley	PA	19547
1025 Ms.	Joan	E.	Coover	348 Snyder School Rd.	Bemville	PA	19506
1026 Ms.	Nancy	J.	Wessner	372 Snyder School Rd.	Bemville	PA	19506
1027 Ms.	Tanya	M.	Feick	1904 Van Reed Rd. apt. G3	Wyomissing	PA	19610
1028 Mr.	Keith		Kaufman	305 Mail Route Rd.	Reading	PA	19608
1029 Ms.	Gloria	F.	Hill	793 Davis Dr.	Reading	PA	19607
1030 Mr.	Thomas	L.	Basinger	RR 1Box 227C	Beech Creek	PA	16822
1031 Ms.	Sabrina	C.	Basinger	RR 1Box 227C	Beech Creek	PA	16822
1032 Ms.	Robyn	D.	Dyke	P.O. Box 560	Milesburg	PA	16853
1033 Mr.	Donnie	S.	Dyke	P.O. Box 560	Milesburg	PA	16853
1034 Ms.	Kimberly	D.	Bookhamer	RD 2 Box 223	Huntingdon	PA	16652
1035Ms.	Nancy	E.	Bunch	498 Zinn's Mill Rd.	Lebanon	PA	17042
1038 Dr.	Earl	F.	Hartlaub	481 Camp Hebron Rd.	Halifax	PA	17032
1037 Ms.	Lori		Tumer	1337 Pine Rd.	Carlisle	PA	17013
1038Mr.	Bruce	L.	Phillips	RR 2 Box 653	Annville	PA	17003
1039Ms.	Janine	M.	Boatwright	RR 1Box 34	Allenwood	PA	17810
1040Ms.	Denise		McHenry-Do	RD 2 Box 571	Annville	PA	17003
1041 Ms.	Amy		Dunigan	2128 Sheri Ln.	Mt. Joy	PA	17522
1042 Mr.	Tom	L.	Morrow	2671 Spring Rd.	Carlisle	PA	17013
1043 Ms.	Melanie	A.	Wertz	17 Woods Dr.	Mechanicsburg	PA	17055
1044 Mr.	Richard	A.	Walker	398 Walker Hollow Rd.	Bellefonte	PA	16823

INSI III MANANTI		IM		中国国际等的企业工程,1975年1980年1980年1980年1980年1980年1980年1980年1980		846	學20世紀
1045 Ms.	Lenore		Steinmetz	1180 Greenleaf Dr.	Bethlehem	PA	18017-9319
1046 Ms.	Karen	L.	Lutz	4 E. First St. P.O. Box 381	Boiling Springs	PA	17007-0381
1047 Ms.	Lisa	L.	Stepler	1012 Shady Beach Rd.	Elkton	MD	21921
1048 Mr.	Jack			RD 2 Box 500	Summerville	PA	15864-9067
1049 Mr.	Paul	R.	lves	802 Kingston Rd.	Govans		21212
1050Ms.	Jean			227 Kimport Ave.	Boalsburg		16827
1051 The Honorable	Scott	E.		162-B East Wing	Harrisburg		17120-2020
1052 Ms.	Debra		Malcei			PA	
1053 Ms.	Jeanne	L.	Price	673 Wiltshire Dr.	State College		16803-1451
1054 Mr.	James			RR 3 Box 408	Muncy		17756
1055 Mr.	David	D.	Maxwell	240 Academy Ave.	Pittsburgh	PA	15228-1250
1056 The Honorable	Matthew	J.	Ryan	Room 139 Main Capitol Bldg.	Harrisburg	PA	17120-2020
1057 Mr.	Michael	D.	Hnatin	935 Louisa St.	Williamsport	PA	17701
1058 Mr.	Roy		Ford	RR 4 Box 2	Shickshinny	PA	18655
1059 Mr.	Waiter		Axsmith	8504 Route 183	Bethel	PA	19507
1060 Ms	Cia	Υ.	Reis	RR 2 Box 336	Muncy	PA	17756
1061 Mr.	Thomas		Yannaccone	415 Elm St.	Watsontown	PA	17777
1062 Mr. & Mrs.	Sidney & Susan		Alpaugh	RD 5 Box 232	Muncy	PA	17756
1063 Ms.	Dorothy		Yannaccone	611 Main St.	Watsontown	PA	17777
1064 Mrs.	Noman		Eaken	29 Appalachian Trail Rd.	Gardners	PA	17324
1065 Ms	Barbara	A.	Aaron	23 Jefferson St.	Warren	PA	16365
1066 Mr.	David		Adams	c/o Ritter Engineering Co.	Zelienople	PA	16063-9698
1067 Ms.	Charlene	1	Wright	P.O. Box 243	Butler	PA	16001
1068 Ms.	Lois	A.	Miller	332 Valencia Rd.	Evans City	PA	16033
1069 Ms.	Joyce	A.	Wyatt	242 McClelland Rd.	Claysville	PA	15323
1070 Mr.	John	Y.	Yanoviak	P.O. Box 678	New Cumberland	PA	17070
1071 Ms.	Kathy		Cawthern	1205 Mahaffey Hollow Rd.	Linden	PA	17744
1072 Ms.	Kimberly	A.	Sachek	850 S. Bridge Rd. Lot 37	Prosperity	PA	15329-1129
1073Mr.	Philip		Sauerlender	245 Ellen Ave.	State College	PA	16801
1074 Ms.	Cherie		Keith	2880 Park Ave.	Washington	PA	15301
1075Mr.	Neal	M.	Lutz	360 Green Valley Rd.	York	PA	17404
1076 Mr.	Russ		Cowles	836 Broad St.	Montoursville	PA	17754
1077 Ms.	Laura		Pittman	17845 White Rd.	Centerville	PA	16404
1078 Mr.	John		Roberts	137 Meyers Rd.	Eighty Four	PA	15330
1079 Mr. & Mrs.	William & Gail		Dillon	46 King Rd.	Cochranville	PA	19330
1080Mr.	Robert	M.	Ross	RR 2 Box 113A	Wellsboro	PA	16901-9616

1 Comments Query 5/19/98

No. Memerine		MI				825	200
1081 Mr.	Alan		Smith	21 Robin Hood Dr.	Irwin	PA	15642
1082 Mr.	Alex		Pappas	RR 8 Box 8128	Stroudsburg	PA	18360
1083 Ms.	Donna	A.	Cays	RR 8 Box 8128	Stroudsburg	PA	18360
1084 Ms.	Joanne	L.	Matthews	P.O. Box 731	Ligonier	PA	15858
1085 Mr.	William	A.	Bennett	RT 66 P.O. Box 397	De Young	PA	16728
1086 Mr. & Mrs.	James & Mary		Greisinger	Box 2	Bakerstown	PA	15007
1087 Ms.	Gail	S.	Pippin	P.O. Box 35	Wagontown	PA	19376-0035
1088 The Honorable	Michael	D.	Marino	Court House	Nomistown	PA	19404-0311
1089 Ms.	Ann	l	Walker	P.O. Box 383	Boalsburg	PA	16827
1090 Ms.	Marie		Karstetter	RR 1 Box 207	Loganton	PA	17747
1091 Mr.	William	C.	Shoemaker	1519 S. Allen St.	State College	PA	16801
1092 Mr. & Mrs.	Ray & Holly		Anderson	RR 2 Box 304	Meshoppen	PA	18630
1093 Mr.	H. Ray		Egolf	P.O. Box 124	Landisburg	PA	17040
1094 Ms.	Jane	M.	Keeney	522 E. Maple St.	Palmyra	PA	17078
1095	IRRC			333 Market Street	Harrisburg	PA	17101

Proposed Rulemaking - State Forests 17 Pa Code Chapter 21

Comments and Responses

Sec. 21.1 Definitions

Comment: The commentator suggested revising the definition of commercial activity

to read, "An activity in which a person, organization or association directly

or indirectly" (Commentator 660)

Response: The Department agrees with this suggestion and has changed the wording

accordingly.

Sec. 21.1 Definitions

Comment: The commentators stated that the definition of Group Activity is unclear

and ask if group activities include groups of hunters, hikers, etc.

(Commentators 22, 28, 632, 651, 850, and 1095)

Response: For an activity to fall under the definition of "group activity," three criteria

must be met. It must be a scheduled event. It must require a large land area or unique land formation, AND it must have the potential to conflict with traditional uses of State Forest land or pose a potential threat to the

environment.

Sec. 21.1 Definitions

Comment: It was suggested that the word "heavy" be deleted from the definition of

pack animal. (Commentator 660)

Response: The Department did not make the suggested change. Eliminating the word

heavy would preclude the use of llamas and dogs with "doggie packs."

Sec. 21.11 Use

Comment: The language of Section 21.11 is complicated and redundant.

Commentator suggested "All waters on State Forest land are open for the

operation of watercraft unless posted as closed." (Commentator 1095)

Response: The Department agrees and has changed the wording accordingly.

Sec. 21.12(a) Mooring and Launching

Comment: Section 21.12(a) appears to prohibit canoe camping after November 15 and

before March 31. The commentator suggested the following language: "Overnight mooring or storage of watercraft, except canoes, kayaks, rafts and other paddle-powered small boats stored overnight at the operator's campsite from November 15 through March 31 is prohibited except on

lands leased from the Commonwealth." (Commentator 660)

Response: It is not the Department's intent to prohibit such camping during those

months. Therefore, the wording of this section has been modified.

Sec. 21.12(b) Mooring and Launching

Comment: Section 21.12(b), in its proposed form, would not include rivers and

streams flowing through State Forest land. The commentator suggested revising the language to include rivers and streams. (Commentator 1095)

Response: The Department agrees. The original word, "waters," has been retained so

that it includes all bodies of water on State Forest land.

Sec. 21.12(b) Mooring and Launching

Comment: Section 21.12(b) should be reworded to make it clear that boats may be

launched without a State Recreation Area Watercraft Permit so long as a valid registration issued by the Fish and Boat Commission is affixed to the

boat. (Commentators 5 and 6)

Response: Subsection (c) of Section 21.12 in the current rules and regulations states,

"A current Fish and Boat Commission number and certificate of watercraft registration permits daily launching only." This section will remain unchanged in the final rulemaking, as indicated by the ellipsis after

Subsection (b) in the proposed rulemaking.

Sec. 21.12(d) (subsection (e) in final rulemaking) Mooring and Launching

Comment: Allowing the launching of small boats, rafts and canoes from any point on

the shoreline will cause erosion of banks. (Commentator 7)

Response: The suggested change has not been made. The Department is of the

opinion that any environmental degradation will be negligible. If such

damage should become evident, that particular point on the shoreline will be posted prohibiting the launching of watercraft.

Sec. 21.12(d) (subsection (e) in final rulemaking) Mooring and Launching

Comment: The commentator suggested adding wording to Section 21.12(d)

(subsection (e) in final rulemaking) to prohibit launching of boats from private docks without the owner's permission. (Commentator 642)

Response: The Department has not added the suggested provision. The Department

does not have the duty or power to enforce private property rights.

Sec. 21.12(e) (subsection (f) in final rulemaking) Mooring and Launching

Comment: Commentator stated that the language in this section is unnecessarily

complicated. Commentator suggested, "Boat mooring is restricted to

designated mooring areas." (Commentator 1095)

Response: The Department agrees and has adopted the suggested wording.

Sec. 21.13 Motorized Boats

Comment: The language in this section is unnecessarily complicated. (Commentator

1095)

Response: The Department agrees and has revised the wording accordingly. ("Only

electric motors may be used for propulsion of motorized watercraft.")

Sec. 21.21(a) Motor Vehicles

Comment: Section 21.21(a) references sections 21.22, 21.23a and 21.23b. However,

Section 21.23b does not exist due to section redesignations.

(Commentator 1095)

Response: The commentator is correct. The reference to Section 21.23b has been

replaced with a reference to Section 21.23.

Sec. 21.21(b)(2) Motor Vehicles

Comment:

The commentator stated that it would be better to post roads as being

closed in Section 21.21(b)(2). (Commentator 10)

Response:

The suggested change has not been made. Most of the Bureau of Forestry's roads that are open to vehicles are already posted as being open. Those that are not will be posted prior to the regulations being adopted. The Department believes that this section will make enforcement much more efficient and will cause the Department's rules and regulations to be identical to the Pennsylvania Game Commission's thereby reducing

confusion among visitors to both agencies' lands.

Sec. 21.21(b)(3) Motor Vehicles

Comment:

Commentator suggested deleting 21.21(b)(3) since it simply restates the

provisions of Section 21.71. (Commentator 1095)

Response:

The Department agrees. Section 21.21(b)(3) has been deleted.

Sec. 21.22(b) Snowmobiles

Comment:

This provision lacks clarity because both the location and the season are in the same sentence. Commentator suggested splitting the proposed provisions into separate subsections or sentences. Commentator also suggests adding the word "only" to the provision concerning location, as follows: "Snowmobiles may be operated only on designated snowmobile

roads " (Commentator 1095)

Response:

The Department agrees with the comment and has adopted the suggested

wording and placed the provision in different subsections.

Sec. 21.23(3) Trail Bikes

Comment:

Paragraph (3) restates the provisions of paragraphs (1) and (2). Therefore

it is redundant. (Commentator 1095)

Response:

The Department agrees. Paragraph (3) has been eliminated.

Sec. 21.23a(b) All-Terrain Vehicles

Commentator suggested adding the word "only" to the provision

concerning location, as follows: "All-terrain vehicles may be operated only

on designated and posted roads " (Commentator 1095)

Response: The Department agrees with the comment and has adopted the suggested

wording. For purposes of clarity, it has also split this provision into a subsection concerning location and a subsection concerning season.

Sec. 21.23a(b) All-Terrain Vehicles

Comment: Section 21.23a(b) should be changed so that ATV's may be operated

through the last full weekend in September rather than until September 24.

(Commentator 1)

Response: The Department agrees with this comment and has changed Section 21.23a

(b) accordingly.

Sec. 21.23a All-Terrain Vehicles

Comment: The commentators proposed banning ATV's from State Forest land.

(Commentators 3, 4, 12)

Response: The Department feels that ATV's are a legitimate recreational pursuit.

Therefore, the Department did not make the suggested change. Section

21.23a addresses the use of ATV's and restricts them to those trails and

areas that have been specifically designated for their use.

Sec. 21.25(b)(1) Parking

Comment: The commentator suggested that a higher fine for parking violations be

included in Section 21.25(b)(1). (Commentators 7 and 182)

Response: The Department did not make the suggested change because the fine is

governed by 18 Pa. C.S. Section 7505 which permits a maximum fine of

\$5.00.

Sec. 21.25 Parking

Comment: The commentator requested that wording be added to Section 21.25 that

would prohibit an able-bodied vehicle operator from parking in designated

handicapped parking areas. (Commentator 23)

Response: The Department agrees with this suggestion. New language has been

added accordingly.

Sec. 21.25 Parking

Comment: The commentator felt that Section 21.25 will severely limit parking on

State Forest land. (Commentator 10)

Response: The Department does not believe this is the case and no change has been

made. Section 21.25 (a) prohibits only parking that hinders the use of the resource by other visitors. This wording mirrors that in the Bureau of

State Parks rules and regulations.

Sec. 21.25 Parking

Comment: The commentator suggested adding a subsection to Section 21.25 that

would allow a State Forest Officer to have an illegally parked vehicle

towed away. (Commentator 651)

Response: The Department did not make the suggested change. State Forest Officers

do not have "police powers" in the traditional sense. Therefore, the police agency that has jurisdiction on State Forest land is the Pennsylvania State Police. If and when certain State Forest Officers are commissioned to have full police powers under the Crimes Code, they will automatically have the

authority to have vehicles towed.

Section 21.26(1) Horses and Pack Animals

Section 21.27(1) Bicycles and Person-Powered Vehicles

Comment: Many commentators expressed concern with the provisions prohibiting use

of horses, pack animals, bicycles and person-powered vehicles on National and State Forest Designated Hiking Trails unless posted as being open to

their use. They felt that these provisions unreasonably favor one

recreational use (hiking) to the disadvantage of other uses. (Commentators 13, 14, 16, 18, 19-21, 25, 27, 29-41, 43-98, 100-103, 105, 106, 108-114, 116-132, 138, 140-143, 151-155, 159, 164, 165, 167, 171, 175, 176, 187-

275, 278, 280, 283, 285, 288, 297, 303-423, 425-627, 629, 630, 632, 633, 638, 641, 647, 653-655, 661, 662, 664, 667-848, 850, 854, 866, 868, 870, 871, 873, 875, 876, 879,895, 898, 912, 914, 917-923, 931, 933, 934, 938, 948-1042, 1044, 1047, 1051, 1053, 1056, 1057, 1060, 1062, 1068-1075, 1077-1079, 1084, 1087, 1088, 1090, 1092, 1093 and 1095)

Many other commentators supported the provisions, claiming that these other uses, particularly horseback riding, damage trails meant for hiking. (Commentators 6, 24, 139, 156-158,161-163, 166, 168-170, 172-174, 177, 182, 184-186, 276, 277, 281, 282, 284, 289-296, 298-302, 628, 631, 634-637, 639, 640, 643, 644, 648, 649, 651, 652, 656-659, 663, 665, 666, 849, 851-853, 855, 856-865, 867, 869, 872, 874, 877, 878, 880, 882-886, 888-894, 896, 897, 899, 901-911, 913, 916, 924-927, 928-930, 932, 935-937, 939-947, 1043, 1045, 1046, 1049, 1050, 1052, 1054, 1055, 1061, 1063, 1065-1067, 1076, 1080, 1082, 1083, 1086, 1089, 1091 and 1094)

Response:

There are two National Hiking Trails and 18 State Forest Designated Hiking Trails in the Commonwealth. The proposed regulations would impose the assumption that these trails would be closed for horses, pack animals, bicycles and person-powered vehicles unless posted as being open. In view of the intense controversy and confusion generated by these proposed regulations, the Department is deleting them from the proposed rulemaking. The Department intends to revisit these issues and will redraft these regulations regarding the use of horses and mountain bikes on State Forest land in consultation with a work group to be made up of interested stakeholders. Meanwhile, the Department retains its authority under Section 21.71 to post trails and other areas as being closed to certain uses.

21.31(1) Prohibitions

Comment:

Section 21.31 (1) prohibits, among other things, the damaging of dead plants without the written authorization of the District Forester. The commentator asked if this section would prohibit having a small campfire made from dead and down wood gathered from the forest floor. (Commentator 6)

Response:

The commentator has a valid point. Therefore, wording has been added to Section 21.31(1) allowing dead wood to be gathered for a small campfire.

Sec. 21.31 Prohibitions

Comment:

Section 21.31 should include a prohibition of timber harvesting during

songbird nesting seasons. (Commentator 28)

Response:

The Department did not make the suggested change. The State Forest Rules and Regulations are designed to govern the conduct of visitors to State Forest land while timber harvesting is controlled by Department policy and coordinated by the Central Office.

Sec. 21.31(1) Prohibitions

Comment:

The commentators suggested including a prohibition of planting exotic, invasive plant species. Further, one of the commentators suggested that paragraph (1) be changed by deleting the words, "berries, nuts, leeks and mushrooms" and substituting the words "edible wild plants or plant parts . . " (Commentators 42 and 1095)

Response:

The Department agrees with these comments and has made the suggested revisions.

Sec. 21.31 Prohibitions

Comment:

The commentators suggested adding a subsection to Section 21.31 that would allow gold prospecting in streams on State Forest land by dredging or the sluice box and shovel method. (Commentators 645, 1058, 1081, 1085)

Response:

The Department did not make the suggested revision. Department policy now restricts recreational gold prospecting to "hand panning" under certain circumstances as addressed in the response to the comment below. Any other method of gold prospecting is at least temporarily destructive to the environment.

Sec. 21.31 Prohibitions

Comment:

The commentator suggested "hand panning" be the only method approved for gold prospecting in streams on State Forest land. He contends that other methods such as dredging and the sluice box and shovel method can be destructive. (Commentator 26)

Response:

The concerns of this commentator are already addressed by Department policy which states as follows: "Recreational, noncommercial hand panning for placer gold or other minerals is permitted in the rivers and streams of State Forest lands except in Wild and Natural Areas and in waters classified as 'Exceptional Value' or 'High Quality.' Any other activity relating to mineral exploration that disturbs the channel bottom, banks, riparian zone or flood plain is prohibited."

Sec. 21.32 Designated Trees

Comment:

Language is complicated. It is suggested that it be changed to "Only

designated trees may be cut or removed." (Commentator 1095)

Response:

The Department agrees. The suggested change has been made.

Sec. 21.33 Cutting Practices

Comment:

The commentator suggested that acreage available for timber harvesting

should be included under Section 21.33. (Commentator 28)

Response:

The Department did not make the suggested change. Timber sales on State

Forest land are tightly controlled by Department policy and the State

Forest Resource Plan.

Sec. 21.34 Removal

Comment:

The commentator suggested adding a set of criteria after Section 21.34 for

the processing of requests to conduct research on State Forest land.

(Commentator 28)

Response:

The Department did not make the suggested change. Research on State

Forest land is already controlled by a set of criteria in the form of policy.

All research requests are reviewed extensively before permission to

conduct research is given.

Sec. 21.61 Camping Permit

Comment:

The first sentence of this section provides a definition of "camping." It

should be moved to Section 21.1 Definitions. (Commentator 1095)

Response:

The Department agrees. The definition has been moved to Section 21.1.

Sec. 21.61 Camping Permit

Comment: The commentator asked if Section 21.61 would preclude someone from

sleeping overnight in their vehicle if they were not using "standard camping

equipment." (Commentator 2)

Response: Section 21.61 would not prohibit someone from sleeping in their vehicle

either overnight or during daylight hours.

Sec. 21.61 Camping Permit

Comment: The commentators objected to Section 21.61 in requiring primitive

backpack campers to obtain a camping permit on the grounds that it is overly restrictive and often difficult to conveniently obtain a permit with short notice. (Commentators 6, 10, 11, 12, 113, 133, 145, 155, 183, 423,

632, 651, 660, 850, 871, 1051, 1054, 1059, 1076 and 1095)

Response: The Department will retain the existing wording in this section so that

backpackers may camp without first obtaining a camping permit.

Sec. 21.61 Camping Permit

Comment: The commentator objected to the Department allowing "motorized

camping" and suggests that no camping permits be issued to campers who

use their vehicles as an integral part of the camping experience.

(Commentator 660)

Response: The Department's Forest Recreation Committee is presently examining this

issue along with other recreation issues. It would be premature at this time

to impose a prohibition against this type of camping.

Sec. 21.62 Open fires

Comment: The commentators suggested adding a provision at the end of Section

21.62 to the effect that this restriction does not apply to small self-

contained camp stoves when being used in a safe manner. (Commentators

115, 1059 and 1095)

Response: The Department agrees with this suggestion and has added the wording.

Sec. 21.62 Open Fires

Comment:

The commentators objected to Section 21.62(b) prohibiting open fires from

March 1 to May 25 and from October 1 to December 1 without

authorization from the District Forester. (Commentators 183, 651, 660

and 1054)

Response:

The Department did not make the suggested change. The spring and fall months are the highest fire danger months of the year. The Bureau of Forestry's field staff has been pointing out the need for this prohibition for many years. The District Forester's authorization for an open fire during these months is easy to obtain. A phone call to a forest district office is all

that is necessary since the authorization need not be in writing.

Sec. 21.63 Hunting and Trapping

Comment:

The commentator suggested replacing the general reference to the

Pennsylvania game laws with a specific reference to the Game and Wildlife

Code. (Commentator 855)

Response:

Since the suggested wording is more accurate, the Department has made

the change.

Sec. 21.63(b) Hunting and Trapping

Comment:

Subsection (b) is redundant and unnecessary. (Commentator 1095)

Response:

The Department agrees. Subsection (b) has been eliminated.

Sec. 21.64(b) Fishing

Comment:

Subsection (b) is redundant and unnecessary. (Commentator 1095)

Response:

The Department agrees. Subsection (b) has been eliminated.

Sec. 21.65 Target Shooting

Comment:

Since most of the commentators felt this section would allow target shooting virtually anywhere on State Forest land the wording in Section 21.65 is apparently subject to too broad an interpretation. (Commentators

10, 99, 104, 107, 133, 136, 137, 144, 160, 286, 287, 881, 887, 927, 1064 and 1095)

Response:

The Department has changed this section to prohibit target shooting except where authorized by the District Forester.

Sec. 21.69 Swimming

Comment:

The deletion of Section 21.69 prohibiting swimming in waters on State Forest land will lead to environmental damage. (Commentator 7)

Response:

The Department has not re-inserted 21.69. Waters in which it would be inappropriate to swim from a safety or environmental point of view will be posted as such thereby prohibiting swimming.

Sec. 21.70a Snow Plowing

Comment:

The commentators felt that Section 21.70a which prohibits snow plowing on State Forest roads without the approval of the District Forester may lead to denying access to property owners. (Commentators 10, 145, 155, 423, 632, 871)

Response:

The suggested change has not been made. The Department does not intend to deny access to any person with a legitimate need. However, it is important that the Department has control over how State Forest roads are plowed. Plowing snow from crowned earthen roads will lead to serious road damage if not done properly. Moreover, many of the State Forest roads are also part of the snowmobile system. It is important that roads be plowed in such a way that this sport is not adversely impacted. However, the wording of the section has been modified so that the District Forester's verbal authorization is sufficient.

Sec.21.71 Closure

Comment:

The commentator suggested that since many provisions of this chapter refer to closure of portions of State Forest land for certain activities or times of year, additional emphasis be placed on this section by moving it to a more prominent location under PRELIMINARY PROVISIONS. (Commentator 1095)

Response:

The Department agrees. This section has been relocated and re-numbered as Section 21.4.

Sec. 21.71(a) Closure

Comment:

The commentator objected to Section 21.71(a) which gives District Foresters the authority to close areas of State Forest land to certain uses without having a monitoring system in place to insure that this authority is

not abused. (Commentator 6)

Response:

The Department made no change to this section. By policy, a monitoring system is already in place.

Sec. 21.74 Disorderly Conduct

Comment:

Paragraph (2) of Section 21.74 should be changed to prohibit the

possession of alcoholic beverages except within leased campsite buildings.

(Commentator 15)

Response:

The Department did not make the suggested change. The legal use of alcohol on State Forest land has seldom been a problem. Most Federal agencies do not prohibit alcohol. Neither does the Game Commission. Alcohol is prohibited from State Forest Picnic Areas by the rules and

regulations for those areas.

Sec. 21.74 Disorderly Conduct

Comment:

The commentators suggested that a provision be added prohibiting loud noise that might disturb other visitors to State Forest land. (Commentators 2 and 1095)

Response:

The Department agrees with this comment and has added Paragraph (3) to

Section 21.74.

Sec. 21.75 Sanitation

Comment:

The commentators suggested that this section is too confusing and too restrictive in its proposed form. (Commentators 42, 651 and 1095))

Response:

The Department agrees with this comment and has re-written the section

accordingly.

Sec. 21.76(a) Feeding of Wildlife

Comment:

Section 21.76 (a) should be changed to allow the summer feeding of

wildlife. (Commentators 4, 8, 424 and 646)

Response:

The Department did not make the suggested change. Comments from the Pennsylvania Game Commission, including one from that agency's chief game biologist, state that artificial feeding of wildlife is unnecessary and

may actually harm wildlife.

Sec. 21.76 Feeding of Wildlife

Comment:

The commentators suggested prohibiting artificial feeding of wildlife, except for songbirds. The commentators felt that artificial feeding of wildlife is unnecessary and possibly harmful to wildlife. (Commentators 424, 650, 855)

Response:

The Department did not make this change. Although the Pennsylvania Game Commission discourages artificial feeding of wildlife, many sportsmen have a tradition of feeding wildlife, especially in the winter months. An outright prohibition of this activity would outrage a large number of people. Instead, with the Game Commission's assistance, the Department will attempt to educate the public concerning the detrimental

aspects of artificial feeding.

Sec. 21.76(a) Feeding of Wildlife

Comment:

The wording in this section is confusing. It should be simplified.

(Commentator 1095)

Response:

The Department agrees. The wording has been simplified.

Sec. 21.76(b) Feeding of Wildlife

Comment:

The commentator suggested that the prohibition be against placing of

feeders that have a single or combined capacity of more than one-half

bushel. (Commentator 26)

Response:

The Department agrees with this comment and has made the suggested

change.

Sec. 21.77 Commercial Activity

Comment:

The commentator suggested that guidelines be included in Section 21.77 concerning how and to whom authorization will be issued. (Commentator 660)

Response:

The Department did not make this change because guidelines should not be a part of the regulatory process. However, the Department's Forest Recreation Committee is in the process of developing guidelines that will be distributed to all District Foresters.

Sec. 21.80(b) Trespass

Comment:

The reference to 18 Pa. C.S. Section 3503 is vague. There are three subsections in 18 Pa. C.S. Section 3503, each of which describes a different type of trespass and degree of offense. Clarification is needed as to the type of trespass and degree of offense referred to in Section 21.80(b). (Commentators 9 and 1095)

Response:

The Department has clarified Section 21.80(b) by specifically referring to subsection (b) of 18 Pa. C.S. Section 3503 (relating to defiant trespasser).

Sec. 21.80 Trespass

Comment:

The commentator objected to this section on the ground that it invites abuse of power. (Commentator 915)

Response:

The Department disagrees. Under the Conservation and Natural Resources Act, the Department has the power and duty to manage, control, protect and regulate the use of State Forest land. The Department, like any landowner, must have the ability to deal with unruly visitors who might cause property damage or who are interfering with the rights of others. The proposed section simply states that a person who engages in this behavior may be ordered by a State Forest officer to leave State Forest land. A person who refuses to leave when ordered commits criminal trespass under Section 3503(b) of the Pennsylvania Crimes Code. The language in the proposed Section 21.80 is identical to that in the Bureau of State Parks rules and regulations.



Pennsylvania Department of Conservation and Natural Resources

Rachel Carson State Office Building, P.O. Box 8767, Harrisburg, PA 17105-8767
Office of the Secretary

October 5, 1998

Mr. Robert E. Nyce Executive Director Independent Regulatory Review Commission 14th Floor, Harristown II Harrisburg, PA 17101

RE: Final Rulemaking - State Forest Rules and Regulations (#7B-1)

Dear Mr. Nyce: Sol,

Enclosure

Enclosed is a copy of a final-form regulation for review by the Independent Regulatory Review Commission pursuant to the Regulatory Review Act.

The Department of Conservation and Natural Resources will provide the Commission with any assistance it may require to facilitate the review of this final regulation. If you have any questions regarding this rulemaking, please contact Glenda Miller, Regulatory Coordinator, at 772-9087.

Sincerely yours,

John C. Oliver

TRANSMITTAL SHEET FOR REGULATIONS SUBJECT TO THE REGULATORY REVIEW ACT

I.D. NUME	BER: 7B-1	63.000 +3 60.40:32
SUBJECT:	State Forest Rules & Regu	lations the state of the state
AGENCY:	Department of Conservat	ion & Natural Resources
	,	TYPE OF REGULATION
	Proposed Regulation	
X	Final Regulation	
	Final Regulation with Notice of P	roposed Rulemaking Omitted
	120-day Emergency Certification	of the Attorney General
	120-day Emergency Certification	of the Governor
	Delivery of Tolled Regulation a With Revisions b Without Revisions	
DATE	FILING SIGNATURE	OF REGULATION DESIGNATION
10/5	Low Comp	HOUSE COMMITTEE ON ENVIRONMENTAL RESOURCES & ENERGY
<u>10/5</u> (Sharty & Mickle	SENATE COMMITTEE ON ENVIRONMENTAL RESOURCES & ENERGY
jojska	Kim C Harner	INDEPENDENT REGULATORY REVIEW COMMISSION
		ATTORNEY GENERAL
September 22	2, 1998	LEGISLATIVE REFERENCE BUREAU